Pre-training in Medical Data: A Survey

Feng Lin¹

Yixuan Qiu¹

Weitong Chen²

Miao Xu¹

¹The University of Queensland, Brisbane 4072, Australia ²The University of Adelaide, Adelaide 5005, Australia

Abstract: Medical data refers to health-related information associated with regular patient care or as part of a clinical trial program. There are many categories of such data, such as clinical imaging data, bio-signal data, electronic health records (EHR), and multi-modality medical data. With the development of deep neural networks in the last decade, the emerging pre-training paradigm has become dominant in that it has significantly improved machine learning methods' performance in a data-limited scenario. In recent years, studies of pre-training in the medical domain have achieved significant progress. To summarize these technology advancements, this work provides a comprehensive survey of recent advances for pre-training on several major types of medical data. In this survey, we summarize a large number of related publications and the existing benchmarking in the medical domain. Especially, the survey briefly describes how some pre-training in many medical scenarios. Moreover, based on the summary of recent pre-training studies, we identify several challenges in this field to provide insights for future studies.

Keywords: Medical data, pre-training, transfer learning, self-supervised learning, medical image data, electrocardiograms (ECG) data.

Citation: Y. Qiu, F. Lin, W. Chen, M. Xu. Pre-training in medical data: A survey. *Machine Intelligence Research*, vol.20, no.2, pp.147–179, 2023. http://doi.org/10.1007/s11633-022-1382-8

1 Introduction

Artificial intelligence (AI) has become a tremendously ubiquitous technique impacting our lives. Applications based on artificial intelligence assist users in making decisions and influencing their daily lives. Technological advances are not possible without the rapid development of deep learning (DL), especially thanks to a much wider adoption of convolutional neural network $(\text{CNN})^{[1, 2]}$, recurrent neural network $(\text{RNN})^{[3, 4]}$, and attention neural network^[5, 6]. Those deep neural networks have been integrated into a variety of research, including several subfields such as computer vision $(\text{CV})^{[7]}$ and natural language processing $(\text{NLP})^{[8]}$.

Medical data analysis is one of the most important sub-filed in AI. The task mainly focuses on processing and analysing the medical data from variant data modalities to extract essential information which aims to help physicians make precise decisions during the diagnosis process. It is anticipated that computer-aided systems will be influential tools in health monitoring and disease diagnosis. A lot of efforts have been successful in current studies, such as processing and analysing medical imaging^[9-11], electronic health records (EHRs)^[12, 13], bio-signals^[14-16] and the data which consists of multiple modalities^[17-20]. Hou et al.^[21-23] utilised CNN to diagnose tumours in the early stages, allowing for early intervention treatment planning to greatly improve the patient's survival rate. A medicine recommendation^[12, 24] was developed as a way to improve patient care by providing personalized recommendations based on electronic health records. Qiu et al.^[14] supported caregivers in identifying cardiac arrhythmias effectively and efficiently, saving more lives. Wang et al.^[17] utilised chest X-rays and the corresponding diagnosis reports training a model for disease diagnosis, similarity search, and image regeneration.

Although existing works have achieved remarkable success, some works found that data-hungry is one of the primary challenges of applying the DNN for processing medical data. On the one hand, some kind of medical data can be obtained easily, but annotating the collected data requires a substantial amount of labour and money; on another hand, in many rare or new disease diagnosis tasks, the data is insufficient because they are too rare to collect or there are issues in privacy. The insufficient data have limited training for a satisfactory model because it could cause overfitting and poor generalization. To address this issue, some large-scale datasets are proposed to make it possible to train satisfactory models. However, the construction of large-scale annotated datasets is labor-consuming and expensive. It is unpractical to develop large-scale annotated datasets.

The researchers, motivated by human learning

 $[\]operatorname{Review}$

Special Issue on Large-scale Pre-training: Data, Models, and Finetuning

Manuscript received June 30, 2022; accepted October 18, 2022; published online February 21, 2023

Recommended by Associate Editor Jun-Zhou Huang ©The Author(s) 2023

strategy, proposed the pre-training to address the issue of lack of annotated data. Considering the human learning strategy, learners can learn a skill based on their prior learning knowledge. For example, learning to play tennis can help in learning badminton.

As summarized in [25], the pre-training technique is specially related to transfer and self-supervised learning. As one of the most critical milestones for solving datahungry issues, transfer learning techniques have explored utilising labelled data and leveraging the unlabelled data effectively. Transfer learning^[26] is a sub-field of machine learning inspired by the process of human learning. It learns the related knowledge in the target domain by transferring information from the same or related domain^[27]. The process of transfer learning consists of two steps, pre-training and fine-tuning. Pre-training is a process of learning universal feature representations and then using the pre-trained model in the downstream tasks, as Fig.1 shows. The recently emerging self-supervised learning is another pre-training paradigm which gets wide notice by more and more researchers. This learning paradigm is committed to extracting abundant knowledge from unlabelled data. Self-supervised learning enables the production of the supervision information by itself instead of manual annotations. In the current studying stage, transfer learning and self-supervised learning are two mainstream pre-training approaches. In this paper, we introduce these two approaches at a high-level and explore pre-training in the medical domain.

1.1 Why pre-training?

However, the emergence of pre-training, mainly including transfer learning and self-supervised learning, provides the opportunity to use a small size of labelled data to train an effective model in the efficient method. In this section, we list the reasons why pre-training is essential. Firstly, the pre-training method was invented from the lack of data information, which is generally divided into the lack of labels and the lack of data volume^[28]. The lack of data volume means that many types of data cannot meet the needs of model training, such as very scarce regional rare disease data. Pre-training can effectively compensate for the impact of this lack of information^[28]. Through pre-training, clusters or potential features in the data are extracted by the model so that the model has more generalization ability for specific content.

Secondly, the utilization of pre-trained models can significantly accelerate the convergence process on downstream tasks. This is particularly beneficial in scenarios where computational resources are constrained.

Thirdly, in the past 20 years, with the rapid development of various industries and the generation of high-performance hardware, a large amount of data has been rapidly generated daily in different industries, such as the medical industry^[29]. However, the cost of manual annotation of datasets increases exponentially. Therefore, the supervised pre-training methods have challenges on the lack of data annotation. Self-supervised pre-training allows us to leverage abundant non-labelled data, getting a good initialization before the downstream tasks.

Also, with the recent advance in self-supervised learning, many studies^[30, 31] have shown that self-supervised pre-training can alleviate the effect of training on data with imbalanced labels.

There are also many applications of pre-training in the medical field. Pre-training technology was first implemented in the medical domain in 2014 by Schlegl et al.^[32], in which they proposed a semi-supervised learning approach to improve lung tissue classification. Specifically, they train the pre-training model with the unsupervised strategy injecting the information from the images without annotation. There are three modalities of the data that we mainly focus on that have been processed with pre-training successfully: medical image data, biosignal, and EHR. Also, the multi-modality scenario has



Fig. 1 Illustration of pre-training. Pre-training is a part of transfer learning. If the pre-training model is fully supervised, the pretraining objectives are required, while if the pre-training model is an unsupervised or self-supervised learning model, the pre-training process does not need the objective.

been considered. For example, the pre-trained BERT model in semantic analysis can be applied to predict future diagnoses using EHR data^[33]. The self-supervised pre-trained model can perform tasks such as classification and segmentation on CT and MRI images^[34]. The electrical bio-signals can be pre-trained to extract the features, thereby helping to perform prediction or diagnosis^[35]. These pre-training applications in the medical field improve the performance of many tasks. Compared with conventional models, pre-training has significantly improved efficiency and accuracy in medical field applications.

1.2 Why is this survey necessary?

There are two reasons why we have organised this survey. First, many works using pre-trained models have achieved satisfactory results in the medical domain in the past few years, but there are few systematic and comprehensive introductions to pre-training models. Second, [25] is a comprehensive survey for pre-training, while there is no such a survey about pre-training in the medical domain. The existing surveys^[36–39] in the medical field only focus on investigating the pre-training models for the specific modality. Particularly, most surveys about pre-training in the medical domain are to review pre-training in medical imaging^[36, 37], and few surveys are published for reviewing processing biosignals^[40] and EHRs^[38]. Therefore, it is significant that we systematically review pretraining approaches in the medical domain.

To the best of our knowledge, this paper is the first systematic and comprehensive summary of the recent pretraining innovations in the medical field, consisting of medical imaging analysis, electric bio-signal data (electroencebhalograms (EEG), electrocardiograms (ECG) etc.), EHRs and multi-modality.

This survey presents the techniques and analysis in a simple manner, which is suitable for a variety of audiences. However, we emphasise the core target audience of the survey mainly for two groups. One group has experts from the medical field who are interested in developing a computer-aided diagnosis system. An additional perspective reader is an expert in machine learning and deep learning and wants to learn about the current developments in pre-training in medicine.

1.3 Collection of paper

We summarize the survey strategy from the bibliographic dataset, keywords of searching conduction, and the main focus on the papers that have been published in conferences/journals.

In this survey, we retrieved papers purely related to pre-training in the medical domain. The related papers retrieving were executed on four well-known bibliography websites, including Google scholar, DBLP, ACM digital library and Web of Science. To collect all the papers possible, we initially searched for the terms "transfer learning/pre-training/self-supervised/contrastive learning" + "medical data/medical images/bio-signal data/EHR/ multi-modality/prognosis". In particular, we pay close attention to top-ranking conferences/journals, including CVPR/ICCV, MICCAI, IJCAI, KDD, ICDM, AAAI, WWW, NeurIPS, ICML, TPAMI, TMI, MIA, Nature, Science, etc. Furthermore, we also screen the results of other conferences, journal papers, and preprint versions on arXiv to ensure that this survey is more comprehensive. We also reviewed many surveys that investigated the pre-training in image processing tasks and NLP-related tasks. As among the collection papers and the former survey, most of them have introduced the basic models, like CNN, RNN, Transformer, self-supervised learning, etc., we will not re-visit these basic techniques and not review the specific papers that introduce the related techniques theoretically. In a particular scenario, model pre-training is usually inextricably linked to the downstream task, including fine-tuning or training a classifier for a particular task.

1.4 Our contributions

This survey aims to present a systematic introduction to recent advances and new frontiers of pre-trainingbased techniques in the medical domain. We summarized more than 200 advanced contributions in this field using pre-training technology, covering the time range from the very beginning of the emergence of pre-training approaches. We list several main contributions of this survey.

1) We first systematically summarized the pre-training techniques that are used for medical and clinical scenarios.

2) We summarized the medical pre-training models used on four main data types: medical images, bio-signal data, EHR data, and multi-modality. To our best knowledge, we are the first to do a survey so comprehensively.

3) We summarized the benchmark dataset of medical images, bio-signal and EHRs.

4) We discuss the challenges of the pre-training model in the medical domain and look to the topics for future research.

The rest of this survey is structured as follows. Section 2 briefly introduces the benchmark datasets in the medical domain and the basic models and methods for pre-training. Section 3 summarises pre-training on medical imaging analysis for different datasets. Section 4 gives an introduction to pre-training for bio-signal. Section 5 summarises the state-of-the-art pre-training methods for EHRs. In Section 6, we discuss the challenges and future directions. Finally, Section 8 gives the conclusion of the survey.

2 Background

This section will summarise the publicly available benchmark dataset in the medical domain. Moreover, some basic pre-train methods are briefly introduced.

2.1 Benchmark datasets

In this section, we extensively explore the benchmark datasets which can be used in machine learning (ML) and DL-based tasks in the medical domain.

2.1.1 Medical imaging benchmarking datasets

Computer vision has been a popular topic in medical imaging processing. There are hundreds of datasets in this field. As listed in Table 1, this study presents a comprehensive overview of 16 frequently utilised publicly accessible medical imaging datasets. Table 1 includes the name of each dataset, the modalities they encompass, their potential applications, and benchmarking results.

2.1.2 Bio-signal medical benchmark datasets

As listed in Table 2, we summarize 22 bio-signal benchmark datasets that are publicly available or have access restrictions. We present the modalities in the dataset, the number of subjects (# Subjects), the number of records (# Records), the sampling rate, the related task, and the comparison of the results.

2.1.3 EHRs benchmark datasets

Based on the survey, we only found four publicly available benchmark datasets are reusable for EHRs related tasks, such as the eICU Collaborative Research Database (eICU)^[41], the Medical Information Mart for Intensive Care III (MIMIC III)^[42], IQVIA¹, and PhisoNet Challenge $2012^{2[43]}$. In some works, they use their private dataset to execute the experiments. In the following, we

Table 1 Statistics of the medical imaging benchmark datasets. Here, SEG is the abbreviation of "Segment", and CLS denotes "Classification". In the column "# of images", the number shows the ratio of number of the training set to number of the testing set.

Dataset name	Modality	# of images	# of classes	Tasks	Benchmarking results $(\%)$	
CHAOS-MRI ^[44]	MRI	992	5	SGE	Dice-socre: 86.75; MSD: $66.00^{[45]}$	
CHAOS-CT ^[44]	CT	2874	2	SEG	Dice-score: 97.79 ± 0.43 ; MSSD: 21.89 $\pm 13.94^{[44]}$	
$NIH-CT-82^{[46]}$	CT	7 141	2	SEG	Dice: $71.80 \pm 10.70^{[46]}$	
$\mathrm{ICCAI}2017\mathrm{LiTS}^{[47]}$	CT	131/70	2	SEG	Lesion: Dice: 82.40 Liver: Dice: $96.50^{[48]}$	
$3 Dircadb^*$	CT	20 venous phase	2	SEG	Tumour: Dice: $93.70 \pm 0.20^{[48]}$	
$OCT2017^{[49]}$	CT	207 130	4	CLS	Acc.: 92.80; Sen: 93.20; Spe: $90.10^{[49]}$	
$\mathrm{BUSI}^{[50]}$	Ultrasound	780	3	CLS	The original tumour image (ROI): Acc: 88.46; F1: 83.33; AUC:90.52 The segmented tumour image: Acc: 90.77; F1:84.21; AUC:88.57 The tumour shape image (TSI): Acc: 85.38; F1: 75.95; AUC:86.60 The fused image: Acc: 94.62; F1: 91.14; AUC:97.11 ^[51]	
$CheXpert^{[52]}$	X-ray	$224\ 316$	14	CLS	Avg. Acc on 14 categories: $90.70^{[52]}$	
ChestXray 14 ^[18]	X-ray	112 120	14	CLS	AUROC: $84.40^{[52]}$	
$EyePACS^{\dagger}$	Fundoscopic	$35\ 126$	5	CLS	Acc: 91.10; AUC: $95.70^{[53]}$	
$ISIC2019^{[54]}$	Dermoscopy	25331	9	CLS	AUC: MEL: 92.80; NV: 96.00; BCC: 94.90; AK: 91.40; BKL: 90.4; DF: 97.9; VASC: 95.60; SCC: 93.80; UNK: $77.5^{[55]}$	
$TCGA-GBM^{[56]}$	WSI	255	-	\mathbf{SA}	C-index: $64.52^{[57]}$	
$TCGA-LGG^{[56]}$	WSI	1 061	-	SA	C-index: 74.10 ^[58]	
$TCGA-LUSC^{[56]}$	WSI	485	-	SA	C-index: $62.87^{[57]}$	
$NLST^{[59]}$	WSI	1 104	-	SA	C-index: 64.76; AUC: 66.93 ^[60]	
PatchCamelyon [‡]	WSI	262144/32768	_	SA	Acc: 89.80; AUC: 96.30; NLL: 26.00 ^[61]	

* https://www.ircad.fr/research/data-sets/liver-segmentation-3d-ircadb-01/

[†] https://www.kaggle.com/c/diabetic-retinopathy-detection/

[‡]https://github.com/basveeling/pcam

 $^1\ {\rm https://www.iqvia.com/insights/the-iqvia-institute}$

² https://physionet.org/content/challenge-2012/1.0.0/

🖉 Springer

Name	Modalities	# Records	# of subjects	Task	Sampling rate	Benchmarking results $(\%)$	
$\begin{array}{c} \text{MIT-BIH arrhythmia} \\ \text{dataset}^{[62]} \end{array}$	ECG	48	1	AD	$360\mathrm{Hz}$	CNN: Acc: 92.56; F1: 92.54; AUC: 99.20 ^[14]	
PTB-XL ^[63]	ECG	21 837	71	AD	$500\mathrm{Hz}$	AUC: 92.90 Deep learning for ECG analysis: benchmarks and insights from PTB-XL	
$\begin{array}{c} \text{MIT-BIH noise stress} \\ \text{test}^{[64]} \end{array}$	ECG	15	1	Denoise	$360\mathrm{Hz}$	CNN: Acc: 96.19; F1: 96.18; AUC: $99.64^{[14]}$	
${f EuropeanST-T}\ {f database}^{[65]}$	ECG	90	2	AD	$250\mathrm{Hz}$	CNN: Acc: 92.53; F1: 91.06; AUC: $99.20^{[14]}$	
AF classification challenge 2017 ^[66]	ECG	8 528	4	AD	$300\mathrm{Hz}$	$F1: 72.00^{[67]}$	
PTB diagnostic ECG ^[68]	ECG	549	9	AD	N/A	Acc: 96.00; Pre: 99.00; Re: 93.00 using 12 leads $^{[69]}$	
$AHA^{[70]}$	ECG	154	8	AD	$250\mathrm{Hz}$	$\text{CNN: ACC: } 99.70; \text{F1: } 99.71; \text{AUC: } 99.98^{[14]}$	
CPSC2018 ^[71]	ECG	$6\ 877$	8	AD	$500\mathrm{Hz}$	$Overall F1: 84.00^{[72]}$	
AMIGOS ^[73]	ECG, GSR	N/A	40	ED	$128\mathrm{Hz}$	$Arounsal(Acc: 76.00); Valence(Acc: 75.00)^{[74]}$	
ASCERTAIN ^[75]	EEG, ECG, EDA	N/A	58	ED	N/A	[75]	
BIO-VID-EMO DB ^[76]	$\mathrm{ECG},\mathrm{EMG},\mathrm{SC}$	N/A	86	ED	N/A	Acc: 79.51 ^[77]	
DEAP ^[78]	$\begin{array}{c} \mathrm{EEG, EDA, EMG,} \\ \mathrm{PPG, EOG, RSP} \end{array}$	N/A	32	ED	$512\mathrm{Hz}$	Arousal(Acc: 77.19, F1:69.01); Valence(Acc: 76.17; 72.43) ^[79]	
DREAMER ^[80]	EEG, ECG	N/A	23	ED	ECG:256Hz	$ \begin{array}{l} \mbox{Valence}(\mbox{Acc: 86.23}); \mbox{Arousal}(\mbox{Acc: 84.54}); \\ \mbox{Dominance}(\mbox{85.02})^{[81]} \end{array} $	
MAHNOB-HCI ^[82]	EEG, ECG, EDA, RSP, SKT	N/A	27	ED	N/A	$M_e(SD_e)$: -3.30(6.88) bpm; RMSE: 7.62 bpm; M_{eRate} : 6.87 ^[83]	
MPED ^[84]	$\begin{array}{c} \mathrm{EEG, ECG, EDA,} \\ \mathrm{RSP, EEG} \end{array}$	N/A	23	ED	N/A	[84]	
$SEED^{[85]}$	EEG	N/A	15	ED	$200\mathrm{Hz}$	Acc: $85.65^{[85]}$	
Temple University Hospital (TUH) ^[86]	EEG	N/A	315	$^{\mathrm{SD}}$	$200\mathrm{Hz}$	Classification error rate: 20.66	
Sleep-EDF: Telemetry ^[87]	EEG, EOG, EMG	N/A	22	SSD	$100\mathrm{Hz}$	Acc: 82.00; MF1: 76.90; $\kappa {:}~76.00^{[88]}$	
$MASS-1^*$	EEG, EOG, EMG	N/A	53	SSD	$256\mathrm{Hz}$	Acc 86.20; MF1: 81.70; $\kappa{:}80.00^{[88]}$	
SHHS ^[89, 90]	EEG, EOG, EMG	N/A	5804	SSD	$12550\mathrm{Hz}$	Pre: 86.00% ; Rec: 87.00 ; Spe: $95.00^{[91]}$	

Table 2	Statistics of the bio-signal benchmark datasets. Here, AD is the abbreviation of arrhythmia diagnosis, SSD denotes sleep-state
	detection, SD denotes seizure detection, ED means emotion detection.

Note: The benchmark results of datasets, ASCERTAIN and MPED, are hard to show in the table. Check them in the original paper. * https://ceams-carsm.ca/en/MASS/

briefly introduce these four datasets.

1) MIMIC-III contains all patients admitted to the intensive care unit (ICU) at Beth Israel Deaconess Medical Center from 2001 to 2012 and includes over 60 000 unique ICU admissions with millions of observations.

2) eICU is a multi-center database comprised of identified health data that contains over 200 000 ICU admissions across the United States between 2014 and 2015.

3) IQVIA is a real-world patient and clinical trial database that can be requested from the website³. This dataset contains 2 609 clinical trials formed between 2014 and 2019, and includes 25 894 doctors across 28 countries.

4) Physionet Challenge 2012 contains 11 records and 988 adult ICU admissions. The dataset is used to predict in-hospital mortality given the first 48 hours of data for each ICU admissions.

2.2 Pre-training

From a historical perspective, the term pre-training was first introduced in 2007 in the works of Bengio et al.^[92, 93] They proposed a model which consisted of greedy layer-wise unsupervised pre-training followed by supervised fine-tuning. The pre-training techniques have been widely used after deep neural networks achieved success^[25].

The research on deep neural networks (DNN) has encountered the bottleneck of a lack of training data (generally lack of annotated data). The problem comes from the

 $^{^3~{\}rm https://www.iqvia.com/insights/the-iqvia-institute}$

fact that it is easy to encounter overfitting and poor generalization issues by training a DNN network with many parameters without sufficient training data^[25]. To solve this issue, in the early stage, some researchers attempted to construct massive annotation datasets for AI tasks from the data level. However, data annotation is a time and cost-consuming task. Moreover, few AI-related tasks have a related large-scale annotated dataset for training, e.g., tasks in the medical domain. How to train a generalized and robust model for specific tasks with few annotated samples has attracted more attention for a long time.

Pre-trained models have achieved significant success in the AI community. We employ those pre-trained models as a backbone to get representative embeddings for downstream tasks. Generally, we adopt the pre-training technique in the following situations.

1) The training and pre-training datasets are related tasks^[25, 94, 95]. If the source and target datasets are similar tasks or in the same domain, we can use the pre-trained model instead of training a model from scratch, which significantly improves efficiency.

2) Training datasets have an extremely small size of the annotated samples^[25, 94]. The small size of the datasets cannot be sufficient to satisfy the training of a highperformance model. A pre-trained model can be introduced under this circumstance to improve the quality of the feature representation, thus getting a satisfactory performance in the downstream tasks.

3) The computation resource is limited^[94]. The pretrained models can speed up convergence on the target task, which allows models to converge sufficiently within fewer iterations. That is friendly for situations where the computation is limited.

4) The data samples are sufficient, but the labelling budget is small^[29]. In the current era of explosive digital data growth, it is easy to collect massive unlabelled data, while annotating them is expensive. In this situation, a self-supervised learning paradigm will help with learning a generalized representation of the unlabelled data and then use such a model to process the downstream tasks.

We list four essential scenarios in which a pre-trained model will be considered to introduce. Though they are mentioned separately, they have overlapped parts, meaning there are no explicit application situations of pretraining. Pre-training can be widely used in multiple circumstances where the rich knowledge benefits various downstream tasks^[25]. As [25] mentioned, pre-training is highly related to transfer learning and self-supervised learning. In the following sections, we will mainly focus on introducing these two paradigms in detail.

The different supervision levels can impact two phases in the pre-training: the pre-training itself and the downstream tasks using the pre-training results. In the pretraining phase, both supervised and unsupervised cases exist. For example, transfer learning could be either supervised or unsupervised, while self-supervised learning can be unsupervised. Whether pre-training methods should be adopted depends on the source data and target data's relationship and the supervision level given in the target data. Generally speaking, the more similarity between the target domain and source domain, the more benefit of the pre-training methods; the less supervision information provided in the downstream task (such as semi-supervised learning), the more benefit will be provided by the pre-training methods.

2.2.1 Transfer learning

Transfer learning is the primary strategy for early pretraining. It is a significant paradigm motivated by the human study process that learners study new knowledge based on previous knowledge. It mainly focuses on solving new problems through the influence of experience and knowledge in pre-training on the target tasks^[25, 96]. This process enables us to use the already learned knowledge from the source domain freely and learn a new skill by accepting little knowledge and speeding up learning. Thus, reflecting the AI task, this process can be generalised to two learning steps: pre-training prior knowledge from the source task and fine-tuning to learn more specific knowledge from the target source^[27]. In transfer learning, pretraining can be categorised into supervised and unsupervised pre-training based on the source data with supervision information or not. Whether the target data is labelled or not, the supervised pre-training can be divided into inductive and transduction transfer learning^[27], as shown in Fig. 2. For transductive transfer learning, a classifier is trained with labelled data, using the pre-trained classifier to produce pseudo labels for the target domain dataset. The data with the produced pseudo labels would be added to the training set. An example of this training fashion is self-training^[97], which is considered a specific semi-supervised learning technique.

Generally, feature transfer and parameter transfer are the two main goals in transfer learning pre-training^[27]. For example, after pre-training a computer vision model on a large labelled dataset "ImageNet", a small amount of data can be fine-tuned to achieve reliable results because the features and parameters of the target task have been learned by the model in the pre-training of the large dataset. In addition, instance transfer and relational knowledge are two other pre-training approaches in transfer learning. Many feasible large-scale models were developed for pre-training, such as AlexNet^[7], VGGNet^[2], ResNet^[98], GoogleNet^[99], DenseNet^[100], etc. In the same way, inspired by transfer learning in the CV domain, pretraining also is widely used in the NLP domain. The pretrained word representation models extract word embeddings as an input of NLP tasks. There have been many



Fig. 2 Illustration of the relationship between pre-training and other methods (unsupervised, supervised, semi-supervised)

well-known pre-training models proposed in recent years, e.g., embeddings from language models (ELMo)^[101] and the well-known pre-trained model BERT^[8] that is trained on a large-scale dataset.

Transfer learning has been so influential in deep learning in recent years that it has become an integral approach to processing medical data. There have been a considerable number of works using transfer learning to improve the performance in medical imaging analysis, such as in radiology^[18, 52], pathology^[102], dermatology^[103], ophthalmology^[104, 105], etc. Most of these previous works were proposed based on a standard pipeline that introduces pre-trained ImageNet models to extract universal representations for various medical imaging modalities. For example, Wang et al.^[18, 52, 106] initialized the weights of backbones from ImageNet pre-trained models. Esteva et al.^[103] demonstrated a DNN-based diagnosis of skin lesions approach using the ImageNet pre-trained model as a feature extractor, getting a competitive performance with dermatologists. Treder et al.^[105] utilised ImageNet pretrained model to extract features for 1112 spectral domain optical coherence tomography images, and Han et al.^[102] utilised the same feature extraction method in the histopathology image classification and segmentation tasks.

Apart from the imaging modality, transfer learning has been successful in other data modalities, but there is no dataset as large as ImageNet for non-image medical data. Some works convert the one-dimensional vector into images with Fourier or Wavelet transformation and then use the ImageNet pre-trained models in feature extraction^[107, 108]. Moreover, other works transfer the feature extractor between different but related tasks^[16, 109]. In addition to these strategies, some researchers pretrained feature extractor on one dataset, which contains relatively large samples, and then transferred the model to process other datasets that could be sparse and small^[16, 110]. Clinical text data is an NLP-related task. Most currently proposed methods use the BERT model to extract word embeddings^[111].

Although transfer learning has successfully processed many tasks, the conventional transfer learning paradigm still has controversial issues in the medical domain. A standard formulation for imaging data is using ImageNet pre-trained models. Matsoukas et al.^[112] pointed out that transfer learning works by increasing the reuse of learned representations. However, there are many remarkable differences in data (including size, distribution, categories, etc.), features, and the final tasks between the natural classification task on ImageNet and the target specification medical data^[95]. He et al.^[94] refered to the fact that</sup> the ImageNet pre-trained model can help to speed up convergence but does not contribute to performance improvement. Especially in relatively large-scale medical datasets, the ImageNet pre-trained model has no obvious advantages compared to a simple model^[95]. In addition, the systematical experiments show that the ImageNet pre-trained model is over-parameterized for medical image tasks instead of extracting more sophisticated fea $tures^{[95]}$.

2.2.2 Self-supervised pre-training

Considering that there are a large number of data produced in the real world that are not annotated, to leverage such data, self-supervised learning has become one of the most promising ways of processing unlabelled data in deep learning^[113–115]. It attempts to gain the supervisory signal from the data pool rather than human annotation, then exploits the underlying semantic information to learn general data representations for downstream tasks.

The typical workflow of self-supervised learning, similar to transfer learning, consists of representation and downstream task learning. The actual self-supervised learning happens in the first stage, where the model learns the knowledge of the unstructured dataset to represent the feature embeddings, which is the exact difference between self-supervised learning and conventional transfer learning. In the downstream learning process, the framework could be the same as in supervised fashion: a

feature extractor followed by a classifier. The top feature extractor will be initialized using the weights transferred from the first stage, and the transferred weights will be fine-tuned as the particular task, training the following modules meanwhile. Another setting is after self-supervised learning without fine-tuning. The extracted features of the unlabelled training dataset will classify using clustering methods, like the k-nearest neighbours algorithm (KNN).

Based on the aforementioned, from the data-driven perspective, the self-supervised learning fashion is similar to the transfer learning settings with unsupervised pretraining: they are all trained on unlabelled data. The selfsupervised learning can be regarded as a branch of transfer learning as a consequence. However, they are trained in different ways: unsupervised pre-training is generally trained on the model without supervision, e.g., clustering^[116]; instead, self-supervised learning typically is trained through an end-to-end framework using the selfproduced supervision information. Furthermore, the selfsupervised learning approach can also be considered semisupervised when the embeddings are fine-tuned with supervision^[117]. The relationship is shown in Fig. 2.

Nowadays, various self-supervised learning frameworks have been developed and succeeded in many domains for applications, such as in the communities of CV and NLP, etc. For instance, many advanced frameworks firstly were developed for CV tasks, like CPC^[118], momentum contrast (MoCo)^[119], SimCLR^[113], BYOL^[114], SwAV^[120], MAE^[121], Siamese^[122], etc. Additionally, BERT^[8] still performs poorly in processing NLP-related tasks. For extended applications, self-supervised learning has become one of the best choices for processing medical data. The reason is that the amount of annotated data is relatively small, while the unlabelled data is considerably large in real-world medical datasets. Therefore, the remainder of this section will delve into the exploration of state-of-the-art self-supervised learning techniques.

To achieve better understanding, we will introduce readers to the state-of-the-art self-supervised learning methods in the upcoming sections.

Contrastive predictive coding (CPC) is an unsupervised contrastive learning approach to learning high-dimensional data representation, which can be used in many data modalities, such as text, speech and image data^[118]. CPC aims to learn useful and informative representations that keep more information on the raw data and can precisely predict the future state latent vector. The architecture of the CPC model has two main components, including a non-linear encoder and an auto-regressive model. The illustrations of CPC frameworks are shown in Figs. 3 and 4, representing the architecture of processing time-series data and images respectively. The non-linear encoder (g_{enc}) maps the raw sequence to a lat-



Architecture of CPC for time-series data Fig. 3



ent representation $(z_t = g_{enc}(x_t))$. The auto-regressive model, like GRU, as a predictor enables the prediction of the context latent representation $(c_t = g_{ar}(z_t))$, which condenses the information before the t state. To evaluate the prediction, here, c_t is used to predict the latent vector of the after-k states. In practice, a linear transformation is used to obtain the predicted latent representation $(\hat{z}_k = W_t c_t)$, where W_t is a learnable linear matrix. A score here defines the relativeness between the predictive representation and the real future sequence. The scoring function can be expressed as the following equation:

$$f_k(x_{t+k}, c_t) = \exp(z_{t+k}^{\mathrm{T}} W_k c_t).$$
 (1)

The non-linear encoder and auto-regressive model are optimised jointly based on the noise-contrastive estimation (NCE) loss, namely, InfoNCE, in this work. The loss function is shown as follows:

$$\mathcal{L}_N = -E \left[\log \frac{f_k(x_{t+k}, c_t)}{\sum_{x_j \in X} f_k(x_j, c_t)} \right]$$
(2)

where the sequence set $X = x_1, x_2, \dots, x_N$ in which has one positive sample, the other N-1 samples are regarded as negative samples^[123].

MoCo^[119] is a mechanism of contrastive learning intuited by dictionary look-up via a dynamic dictionary with a queue and moving-averaged encoder, as Fig.5 shows. MoCo learns the robust representation via performing dictionary look-up by making the query embeddings closed and its matching key embedding far away. There are two necessities to build such a reasonable dictionary: 1) The dictionary should be large enough; 2) The representation should be consistent (the key requires to be encoded using a similar or identical encoder to be meaningful of the similarity metric between query and key in the dictionary). The proposed MoCo mechanism achieves the first necessity through a dynamic memory bank, where the oldest mini-batch will be progressively replaced by a new one. At the same time, a momentum update method is introduced to address the key representations' consistency, as defined in (4) defined. The model will be optimized in MoCo with the InfoNCE loss, which is defined as the following equation:

$$\mathcal{L}_{q} = -\log \frac{\exp(f_{q}(x^{q})f_{k_{+}}(x^{k_{+}})/\tau)}{\sum_{i=0}^{K}\exp(f_{q}(x^{q})f_{k_{i}}(x^{k_{i}})/\tau)}$$
(3)

where τ is a temperature hyper-parameter that is used to smooth the loss, $f_q(x^q)$ denotes the query that is the encoded features and a set of embeddings $\{k_0, k_1, k_2, \dots\}$ encoded with $f(\cdot)$. The encoded samples k_s are the keys of a dictionary in which a single key k_+ matches q, and the k_i is considered as the negative sample for q. Another important technique, the momentum update, is defined as

$$\theta_k \simeq m\theta_k + (1-m)\theta_q \tag{4}$$

where θ_k is the weight of the encoder $f_k(\cdot)$ and θ_q is regarded as $f_q(\cdot)$. $m \in [0, 1)$ is a momentum coefficient. A larger m yields better performance, as the experiments show in [119]; when m = 0.999, the performance is best. In further research, the authors of MoCo proposed MoCov2^[124] and MoCo-v3^[125] to improve the performance.

Swapping assignments between multiple views (SwAV)^[120] is a cluster assignment-based contrastive learning paradigm. The illustration of SwAV is shown in Fig.6. Compared to the previous contrastive learning methods, SwAV does not calculate view-pair comparison, instead comparing the cluster assignments of the different views, thus not requiring huge computation resources. Apart from introducing a clustering mechanism, SwAV proposes a multi-crop augmentation strategy aiming to increase the number of image views while not burdening with extra memory and computation cost. Piratically, SwAV introduces online clustering, which maps the features $(Z = z_1, z_2, \cdots, z_B)$ to a set of vectors $(Q = q_1,$ q_2, \cdots, q_B) by the prototype $(C = c_1, c_2, \cdots, c_K)$. Then, the newly defined loss is to minimize the similarity with a setup of swapped prediction. The loss shows as the following equation:

$$\mathcal{L}(z_t, z_s) = l(z_t, q_s) + l(z_s, q_t) \tag{5}$$

where the $l(\cdot)$ is described as the following equation. Here, we give an example with $l(z_t, q_s)$.

$$\ell\left(oldsymbol{z}_t,oldsymbol{q}_s
ight) = -\sum_k oldsymbol{q}_s^{(k)}\logoldsymbol{p}_t^{(k)}$$

where



Deringer

156

$$\boldsymbol{p}_{t}^{(k)} = \frac{\exp\left(\frac{1}{\tau}\boldsymbol{z}_{t}^{\mathrm{T}}\boldsymbol{c}_{k}\right)}{\sum_{k'}\exp\left(\frac{1}{\tau}\boldsymbol{z}_{t}^{\mathrm{T}}\boldsymbol{c}_{k'}\right)}.$$
(6)

Simple framework for contrastive learning of visual representations (SimCLR)^[113] is a straightforward implemented framework for contrastive learning, which was first proposed to process image data, yielding state-of-theart performance. The framework of SimCLR is shown in Fig. 7. It learns representation on an unlabelled dataset by maximizing agreement between random conducting augmentation methods for the same data sample via a contrastive loss. Similarly, the same data was augmented with two randomly selected methods producing two views. These two views are treated as a cheerful pair, while considering all other samples in the same batch as the negative samples. The augmented views pass through the backbone, typically a large-scale neural network, producing the embeddings that are the features of the data we want to get via the pre-training settings. The dimensionality of trained embeddings is reduced through a multi-layer non-linear projection head. The losses are calculated by conducting the loss function for a cheerful pair and its corresponding negative samples in a mini-batch.

The contrastive loss function is defined as

$$\mathcal{L}_{CL,i} = -\log \frac{\exp(s_{i,j}^{+}/\tau)}{\exp(s_{i,j}^{+}/\tau) + \sum Q_{[i \neq k]} \exp(s_{i,k}^{-}/\tau)}$$
(7)

where s^+ is the matrix of the similarity between one positive pair (z_i, z_j) , s^- is the matrix of the similarity between the negative pairs (z_i, z_k) , τ represents the temperature parameter used to smooth the labels, and Qmeans the distribution of the mini-batch in the current state. The similarity matrix could be calculated with $sim(u, v) = \mathbf{u}^{\mathrm{T}} \mathbf{v} / ||\mathbf{u}|| ||\mathbf{v}||.$

Bootstrap your own latent $(BYOL)^{[114]}$ is another popular approach to self-supervised representation learning. The framework of BYOL is shown in Fig.8. Unlike the SimCLR and MoCo^[119], BYOL can learn high-quality representations relying on only one augmented view of an image, which means that it does not require negative samples. Particularly, the framework consists of two neural networks to learn the representation, where an online network is trained to predict the representations produced by the target network. In this way, the additional predictor module in the online network can prevent the model's collapse. The loss function can be defined as the following mean square error (MSE) between the predic-



tion from the predictor of the online network and the output of the target network projection:

$$\mathcal{L}_{\theta,\xi} \triangleq \|\bar{q_{\theta}}(z_{\theta}) - \bar{z'_{\xi}}\|_{2}^{2} = 2 - 2\frac{\langle q_{\theta}(z_{\theta}), z'_{\xi} \rangle}{\|q_{\theta}(z_{\theta})\|_{2}\|z'_{\xi}\|_{2}} \qquad (8)$$

where the z_{θ} is a representation $(y_{\theta} = f_{\theta}(v))$, where v is the augmented sample, and f denotes the encoder) passing through the projection, $z_{\theta} = g_{\theta}(y_{\theta})$; $q_{\theta}(z_{\theta})$ is the prediction in the online network; z'_{ξ} is the projection of the target network; and $\|\cdot\|_2$ denotes as the l_2 normalization. When the parameters θ , ξ are updated, the stochastic optimization step will minimize the total loss of BYOL, $\mathcal{L}_{\theta,\xi}^{BYOL} = \mathcal{L}_{\theta,\xi} + \widetilde{\mathcal{L}}_{\theta,\xi}$ where $\widetilde{\mathcal{L}}_{\theta,\xi}$ represents the augmented sample v feeding into the target network while the other augmented sample v' enters the online network.

Simple Siamese networks (SimSiam)^[122] proposed a hypothesis on the implication of the stop-gradient, which plays an indispensable role in preventing collapsing effectively, as shown in Fig.9. Practically, it is the same as SimCLR, BYOL, SwAV, etc., which all utilise the Siamese network, that a single sample is augmented to two views and then processed by the same encoder network. The difference is that on one side, the encoder is followed by a predicted MLP, while the other side has no MLP but with a stop-gradient operation. The model is optimized by maximizing the similarity between the outputs of the predictor and the encoder. Concretely, we can formulate the process to the outputs of $p_1 = h(f_{\theta}(A(x_1)))$ and $z_2 = f_{\theta}(A(x_2))$; minimizing their negative cosine similarity with the following equation:

$$\mathcal{D}(p_1, z_2) = -\frac{p_1}{\|p_1\|_2} \frac{z_2}{\|z_2\|_2} \tag{9}$$

where $\|\cdot\|$ represents the l_2 -norm. The final loss can be defined as a symmetrical loss with a stop-gradient operation (sg) as

$$\mathcal{L} = \frac{1}{2}\mathcal{D}(p_1, sg(z_2)) + \frac{1}{2}\mathcal{D}(p_2, sg(z_1))$$
(10)

where $p_2 = h(f(A(x_2)))$ and $z_1 = f(A(x_1))$.

Masked autoencoders (MAE) are self-supervised learners that randomly mask patches of images and predict the missing pixels^[121]. The architecture of MAE is shown in Fig. 10. MAE employs the random masking strategy. Specifically, MAE randomly samples and masks a portion of patches from the images based on a uniform distribution. Each masked patch is replaced by a token, a shared and learnable vector. In the MAE, the encoder is a ViT model that is only used to process visible patches to



Deringer

obtain its embeddings; the decoder is a light model built with several transformer blocks, and the last layer of the decoder is an MLP. The dimension of the MLP module's output is the same as the patches, which is used to predict the pixels of the masked patches. The encoder inputs are masked tokens and encoded visible patches combined with positional embeddings. Finally, a simple MSE loss will be introduced to calculate the value of the loss between the predicted and original pixels.

BERT^[8] stands for bidirectional encoder representations from Transformers, the illustration shown in Fig. 11. Based on the investigation, BERT profoundly affects the processing and generation of EHRs data and textual medical data in the medical field^[126]. The BERT model adopts the main structure of the bidirectional deep Transformer, which will be introduced next.

The Transformer is an attention mechanism that uses the structure of the encoder and decoder to calculate the relationship between input information^[127]. After the input is passed through the encoder, the contribution of an input element to the total input can be calculated^[127]. In natural language processing (NLP), this attention score is used as the weight of other words for that word to compute a weighted representation of a given $word^{[128]}$. The influence representation of a given word can be obtained by feeding a weighted average of all word representations into a fully connected network. When passing through the decoder, only a one-word representation can be decoded in one direction at a time, and each decoding step will consider the previous decoding results^[127]. After the birth of Transformer, the development of large-scale selfsupervised language generation in the field of NLP is improved remarkably.

After pre-training, BERT can obtain robust parameters for downstream tasks. By modifying inputs and outputs with data from downstream tasks, BERT can be fine-tuned for any NLP task^[128]. BERT can handle these applications efficiently by inputting a single sentence or sentence pair. For input, its pattern is two sentences connected by a particular token [SEP], which can repre $sent^{[128]}$:

1) Sentence pairs in paraphrasing;

2) Hypothesis-premise pairs in implication;

3) Questions in question answering-paragraph pairs;

4) Single sentences for text classification or sequence tagging.

For output, BERT will generate a token-level representation for each token, which can be used to process sequence labelling or question answering, and unique tokens [CLS] can be fed into an additional layer for classification^[128].

In this section, we provide a high-level introduction to benchmark datasets in the medical domain and representative pre-training strategies, as this paper focuses on pretraining in the medical domain, which will make the readers who are not specialised in pre-training techniques quickly and clearly learn about the developments of the related methods and the latest techniques.

3 Medical images in pre-training

The CV technique has been widely used in medical imaging, providing excellent technical support for clinical tasks^[129]. In the fold of medical imaging, three main tracks are receiving more attention, such as diagnosis, segmentation, and survival prediction. The image data modalities include CT^[130], MRI^[131], X-rav^[132], Ultrasound^[133], Dermoscopy^[55], Ophthalmology^[134], whole slide tissue images (WSI)^[60], etc. In recent years, learning in medical images has changed from traditional heuristic learning to learning-based learning, which means that new learning methods can obtain essential information from a large number of unlabelled medical images^[135].</sup> The information in medical pictures can either be marked manually or extracted by the mechanism of a deep learning network. The manual annotation of datasets with over millions of samples is very expensive, and the privacy of medical information is also essential because the information of many medical image data is not shared, especially some particular disease images^[136]. In terms of that, the concept of transfer learning has been considered



Fig. 11 Illustration of BERT^[8]

with a smaller number of labelled images. Some ImageNet pre-trained models are used in processing medical imaging tasks. However, in practice, ImageNet pre-trained models are not compatible with the downstream task of medical images, so self-supervised learning is also developing rapidly in the field of medical images^[95]. Theoretically, a pre-trained model used in medical tasks not only significantly reduces the labour cost of data processing but also improves the efficiency of the model learning process. The investigation of current pre-trained models in medical images will be classified by diverse medical tasks.

3.1 Diagnosis

In discussing pre-training of medical images for the diagnosis, the classification characteristics of brain tumours are mainly investigated because of the large number of studies on this type of data set.

3.1.1 CT/MRI

Early diagnosis is crucial for treating brain tumours; however, separating the MR effects of the brain into the tumour and normal processes is a time-consuming task. Reference [137] is an early transfer learning method using pre-trained brain tumour recognition, using the large natural image dataset ImageNet, and achieved 81% accuracy in the leave one out cross-validation method. Reference [130] is an early article on extracting lung tumour data features through CT slice pre-training. It investigates using pre-training methods to improve models that predict long-term and short-term survival probability. Prakash and Kumari^[138] proposed a method for transfer learning. Using VGG16 and ResNet pre-trained on ImageNet, combined with magnetic resonance brain images from the Harvard Medical School database, the final result can reach 100% classification accuracy. Compared with the previous classification results of ordinary CNN, the pre-training of transfer learning combined with the downstream task technology of image enhancement improves the classification accuracy. Khan et al.^[139] proposed a data-augmented pre-training method, compared to previous VGG16 and ResNet, the designed CNN model is applied on a small MRI brain slice dataset for brain tumour detection, and this method is faster and more accurate. Based on the previous augmented data treatment, Sajjad et al.^[140] used a CNN model to segment the tumour region, used the data augmentation mode to expand the segmented data for pre-training and then finetunes the pre-trained CNN model. The article uses many brain tumour data sets to compare the model results. Compared with the tumour grading results before data augmentation, the transfer learning pre-trained model after data augmentation has higher accuracy.

Deepak et al.^[141, 142] extract the features of brain tumours in the data set in the pre-training of transfer learning and simulate the size of different pre-training data in the experiment. The data set is taken from [143] on brain tumours. In [141], the authors employed Support Vector Machine (SVM) and KNN classifiers in the downstream tasks. The results indicate the exceptional capability of the pre-trained model to extract valuable features from brain MRI images. The integration of the pre-trained model enhances robustness and yields superior performance with a small amount of training data. The data set will have relatively good results after testing, but the recognition of brain tumours type meningioma is not as good as the other two.

Small-scale tumour image datasets are common, so there are some investigations on the pre-trained methods to mitigate the influences of the insufficient scale of image data. Swati et al.^[144] used transfer learning to perform brain tumour classification studies on brain magnetic resonance images, pre-training on small-scale data, and the dataset is [145]. The experimental results on this dataset are state-of-the-art on a small scale of data. Wang et al.^[146] discussed the use of ResNet to pre-train on the public Luna16 dataset and then fine-tune it on the lung cancer data of Shandong Provincial Hospital, with an accuracy rate of 85.71%, which is better than the existing AlexNet, VGG16 and DenseNet on lung cancer. Also, in small (100 sample lung CT images) training, a high level of performance can still be maintained.

Moreover, many novel pre-trained structures in medical image diagnosis appeared with reliable performance. Marentakis et al.^[147] investigated the classification of CT images of non-small cell lung cancer (including adenocarcinoma and squamous cell carcinoma) and compares it to four types of technical models. It is found that the structure of long short-term memory (LSTM) + Inception performs the best, which is better than experts' classification accuracy, which is 7%-25% higher. Because of LSTM, this model is not allowed to perform segmentation on the image, so the technique is not affected by differences in the edges of the image. Kutlu and Avci^[148] established a new model to classify liver and brain tumours. First, the pre-trained model of the AlexNet architecture is used to extract features from the input data, and then the features of wavelet transform can be used to extract essential factors to improve the classifier's performance. Finally, LSTM has the function of the signal classification.

3.1.2 Ultrasound

Given that current radiologists' professional skills and knowledge are not very reliable, many abnormal ultrasound images of fatty liver cannot be well diagnosed, leading to the development of fatty liver into a fatal chronic disease. In order to improve the accuracy of ultrasound image classification, Reddy et al.^[133, 149] proposed the convolutional neural network combined with transfer learning (VGG16 pre-train) to analyse and identify whether there is fatty liver. At the same time, these two articles compare the ordinary CNN without pre-training and other non-deep learning methods. The results show that the pre-training and fine-tuning of transfer learning have significantly improved the recognition rate, both of which remain above 95%.

3.1.3 X-ray

Benign and malignant breast tumours are difficult to distinguish under X-rays. In the recent popular transfer learning, most pre-trained models are trained on the mainstream ImageNet benchmark datasets. Since these datasets do not contain breast-related images, the recognition results are not ideal. Alkhaleefah et al.^[150] proposed a double-shot transfer learning model. The pretraining uses the image enhancement mode to reduce the problem of over-fitting and insufficient data. Compared with other mainstream pre-training models, the recognition accuracy is greatly improved. In [132], two models are designed to improve the recognition of lung diseases by X-ray and CT, respectively: one is an improved AlexNet, and the other is a combination of human operation and pre-trained learned features to improve the classification accuracy. The improved pre-training method improves the accuracy by about 10% compared to the original training method. The two training datasets are [151] and [152].

3.2 Segmentation

3.2.1 Abdominal organ segmentation

The abdomen is a vital part of the area in the human body, referring to the region between the thorax and pelvis^[27]. Abdominal organ segmentation has significance for reducing patients' mortality. In the task of abdominal organ segmentation, single or multiple abdominal organs are segmented into semantic segments of pixels identified with homogeneous features, such as colour and texture^[27]. Automatic segmentation of lesions on liver images is an essential step for correct decision-making in clinical diagnosis. In [153], a cascade fully convolutional neural network is proposed to automatically segment the liver and lesions in CT and MRI abdominal images. The first convolutional network was used to identify the location of the liver in the abdominal picture, and the second network was used to identify the lesion site. Both convolutional neural networks are pre-trained using U-Net^[154]. The accuracy of the experimental results of Dice has reached the current best level, and the model can be finetuned to adapt to different situations. Conze et al.^[155] designd a multi-organ segmenter for CT and MRI images of the abdomen and extends standard conditional generative adversarial networks. At the same time, Cascade's pre-trained encoder-decoder structure extracts the features of organs and identifies abdominal organs through contextual information. The adversarial generative network in the paper achieves better segmentation than the encoder-decoder structure. In addition, Kavur et al.^[44] introduced a medical image segmentation competition held

at CHAOS dataset, in which players use the prestained encoder-decoder model for training for single or multiple organ segmentation tasks. In unimodal and multi-modal tasks, pre-trained deep learning models show comparative advantage results compared to other methods.

The training of 3D images is not efficient. Therefore, Li et al.^[48] proposed a method using a 2D UNet (H-DenseUNet) and a corresponding 3D DenseUNet for liver cancer segmentation computation. The convergence speed of the pre-trained transfer learning is significantly higher than that of the ordinary model. This method was evaluated on the MICCAI 2017 liver tumour challenge^[47] and the 3DIRCADb dataset^[156], respectively, and also achieved state-of-the-art results. The application of 2D and 3D image models has merged algorithms with time advantage and memory space advantage.

3.2.2 Ultrasound

Convolutional neural networks have shown promising results in breast tumour segmentation in ultrasound. Generally, these CNN-based methods modify the architecture model or use the CNN ensemble to design new models. Gómez-Flores and Pereira^[157] evaluate the segmentation of breast tumour ultrasound images using four transfer learning models, including AlexNet, U-Net, VGG16 and VGG19's SegNet, and ResNet representations. These pre-trained models are fine-tuned on normal and tumour breast images, where the datasets come from [50, 158]. In these ultrasound breast-specific datasets, the F1 value of the test after pre-training on ResNet18 is the highest, which indicates it has more potential capability. Similarly, in [159–161], investigations of the segmentation of ultrasound breast cancer with transfer learning are included, and [162] also made the comparison of kidney image segmentation with pre-trained methods. In contrast, these investigations prove that pre-training can be efficiently and precisely used for ultrasound image feature extraction.

3.2.3 Comprehensive

The application of self-supervised learning in segmentation is also pervasive. Based on many basic computer vision models, especially the method of exchanging segmentation positions, the pre-trained encoder can accurately learn the features of the picture. Bai et al.^[163] used U-Net to pre-train and test the accuracy of the training set of tiny hearts, and the experimental result improves the accuracy by about 0.04 compared with the ordinary U-Net. Li et al.^[164] conducted a pre-training experiment on image rotation, which is also a popular model for selfsupervised learning. The method is similar to SimCLR's image enhancement, derived from the model relative positions of image patches^[165, 166]. This method performs pseudo-label classification for clusters in the results of self-supervised learning. Experiments show that this pseudo-label pre-training method reduces labour costs by 80% and achieves the same level of segmentation accuracy. The pre-training of self-supervised learning mainly does not have annotated requirements for the input information, and the data characteristics of the pictures learned under the encoders of different pre-training methods are more similar to the intrinsic data features. Chen et al.^[167] proposed a method to learn semantic features of medical images using self-supervised learning while investigating the effect of unlabelled pre-training applications for classification, localization, and segmentation. Semantic features can be appropriately learned by the context restoration method, and the results of various pretrain scenarios prove that self-supervised learning has reliable performance on medical image tasks.

3.3 Survival prediction

As well as classification and segmentation tasks have received much attention, survival analysis also plays a critical role in current clinical practice as a part of computer-aided medical image analysis. Survival prediction (survival analysis or prognosis) is a medical task to predict the expected duration of time until events happen (e.g., death), which is frequently used for cancer patients^[60]. Some works have utilised deep learning methods to achieve state-of-the-art survival prediction results^[168, 169]. However, the requirement for large amounts of well-phenotyped training data has still been one of the significant challenges for introducing deep learning into survival prediction^[170]. There are very few large, labelled, and public datasets. It may be possible to overcome the challenge of limited data by pre-training on a large dataset from another domain^[170]. Therefore, some</sup> works introduce pre-trained models or pre-trained strategies. Li et al.^[171] considered in the survival prediction that the interesting event may not be observed during the study period, and collecting sufficient annotated training samples for robust prediction is extremely difficult in real practice. A transfer learning-based Cox method, namely Transfer-Cox, was proposed to use auxiliary data in a situation where the training data is insufficient. This method aims to extract valuable knowledge from the source domain and transfer it to the target domain with the $L_{1,2}$ -norm penalty for learning a shared representation across the source and target domain. Agravat and Raval^[172] demonstrated a CNN architecture for glioma segmentation and feature extraction, and the extracted features are used to predict the survival of patients with random forest regression. To reduce the impact of high imbalance in the brain tumour segmentation task, in the initial stage, the network is trained for the whole tumour, which provides tumour localization in the brain, and in the next stage, the parameter of the network in the first stage will transfer to process sub-component (e.g., oedema, enhancing tumour and necrotic core). Yao et al.^[60] developed the deep attention multiple instance survival learning (DeepAttnMISL) model to predict cancer survival accurately. For the feature extraction process, they used an ImageNet pre-trained VggNet to extract features from image patches, and Setio et al.^[173] found that using medical pre-trained models positively impacts survival tests for two survival prediction approaches, DeepAttn-MISL and WSISA^[169]. Chen et al.^[174] proposed the hierarchical image pyramid transformer (HIPT), two stages self-supervised pre-training framework to leverage the natural hierarchical structure inherent in WSIs to learn high-resolution image representation for cancer sub-typing and survival prediction. The self-supervision part of this work uses student-teacher knowledge distillation (DINO), where one of the paths in Siamese is the teacher network and another is the student network.

3.4 Longitudinal images data

The images captured from the same area at different times can be considered time-series data. For instance, the CT images of the same body area scan at 1, 3 and 6 months, respectively^[175]. The set of images can be regarded as time-series images data that contain abundant temporal relevant diagnostic information. Integrating temporal information into medical imaging learning has significance for enhancing the diagnosis, prognosis, and disease progression analysis^[175–177]. Some previous works used the CNN and recurrent neural network to mine the temporal and spatial information simultaneously^[176, 178]. However, with our investigation, only a few works are involved in employing a pre-training approach in this field. Xu et al.^[175] demonstrated using the deep learning method to predict prognostic endpoints of patients treated with radiation on longitudinal CT imaging obtained follow-up. In this work, they use ImageNet pre-trained model to extract CT image features. Ouyang et al.^[179] proposed a longitudinal neighborhood embedding (LNE) to capture the gradual deterioration of brain structure and function caused by ageing. They construct a smooth trajectory field that is built by graph construction in each training iteration in the latent space to capture the global morphology while maintaining the local continuity. The extensive experiments demonstrate that the LNE is positive for exploiting the association of information temporal and spatial to reveal the impact of neurodegenerative disorders. Ren et al.^[131] presented a local and multiscale spatial-temporal representation learning method for pre-training on longitudinal MRI imaging datasets, while they proposed various regularisations for avoiding collapsing when extending to multi-scale spatial-temporal representations. They evaluated the improvement in longitudinal neurodegenerative adult MRI and developing infant's brain MRI for segmentation tasks. Konwer et al.^[177] proposed a framework to improve clinical prediction tasks using limited temporal medical images. The proposed framework consists of two modules: temporal progression learning and snapshot learning. The temporal progression learning extracts temporal image sequences using a temporal ConvNet and a self-attention module. Snapshot learning includes self-supervised learning on unlabelled data and then using the target data to fine-tune the network. A re-calibration network is utilised to align these two contextual representations. The experiments demonstrate that this framework outperforms other advanced clinical prediction methods.

3.5 Section conclusion

In all, the main progress of medical images comes from the new field proposed by computer vision, and the impact of pre-training on traditional machine learning and deep learning is huge. Table 3 illustrates the major papers discussed in this section. Transfer learning and self-supervised learning solve the problem of image labelling and the problem of fewer data in pre-training, and the accuracy of pre-training segmentation and diagnosis can generally achieve more accurate results than traditional supervised learning. The application of pre-training on pictures greatly improves the function of auxiliary medical detection, reduces the workload of doctors, and improves the reliability of diagnosis.

4 Bio-signal data in pre-training

It is known that there are several different types of bio-signal data in the medical domain, such as electroencephalograms (EEG), electrocardiograms (ECG), heart rate variability (HRV), electromyograms (EMG), electrodermal activity (EDA) and photoplethysmography (PPG), of which contain a large volume of physiological information. DL-based advances in bio-signals enable the processing of signals (signal segmentation, wave detection, and noise removal) and the creation of high-quality feature representations to be used in clinical applications, such as signal de-noising, disease diagnosis, emotion recognition, genetic mutation detection, etc. EEG and ECG are representative bio-signal data, and there are many publicly available datasets, most of which are generally annotated. Therefore, we collect many pre-trained related papers based on EEG and ECG signals, not to say that the other medical time-series data is unimportant.

Despite a massive breakthrough in algorithms and the increasing availability of publicly available datasets, the lack of annotated data continues to pose one of the most significant challenges to developing bio-signal processing in artificial intelligence. Some researchers have been using pre-training model techniques on many different tasks to address the data scarcity issue. In the remainder of this section, we summarise the current state-of-the-art research on using pre-training methods to process bio-signal datasets based on tasks of all kinds based on bio-signal datasets.

4.1 Pre-processing

Pre-processing the raw signals is one of the essential

 Table 3
 Summary of pre-training related publications with diverse image types in the medical field

Tasks	Image type	Papers		
	CT/MR	[130, 137 - 142, 144, 146 - 148]		
Diagnose (Classify)	X-ray	[133, 149]		
	Ultrasound	[132, 150]		
Segmentation	Abdnomial	[44, 48, 153, 155]		
	Ultrasound	[157, 159-161]		
	Comprehensive	[163, 164, 167]		
Survival WSI prediction		[58, 60, 171, 172, 174]		

steps for bio-signal-related research. The raw data contain multiple noises that are probably caused by different factors. Antczak^[180, 181] proposed an approach that uses the pre-trained model to remove the noise from the raw ECG data; notably, they pre-trained the model with the synthetic data and fine-tuned the real data to create a state-of-the-art noise-removing neural network. In addition, QRS wave detection is an essential task for ECG prepossessing. Rodrigues and Couto^[182] utilised transfer learning to detect the QRS wave and predict the next QRS wave and the shape of the next ECG segment.

4.2 Disease diagnosis

Disease diagnosis with bio-signal data has a vast range of applications since it has been well-studied in various scenarios. Specifically, tasks include arrhythmia diagnosis, atrial fibrillation diagnosis, epileptic seizures detection, etc. Pathak et al.^[166, 183] attempted to use the pre-training method to develop an automatic arrhythmia diagnosis system on one dataset and fine-tuning it on another dataset to evaluate the effectiveness of the pre-training model for ECG data, in which the data used in the tasks were all labelled by cardiovascular experts and the dataset used for pre-training and the fine-tuning under the same tasks. The works^[16, 184, 185] employed the same method to detect atrial fibrillation (AF), but they trained the model on a general ECG dataset and fine-tuned it on an AF dataset. In addition, this type of transfer learning can be applied to EEG datasets to diagnose epileptic seizures in a similar way^[200, 201], which all utilise the CNN network, and Raghu et al.^[200] converted the EEG signals into images using short-time Fourier transforms (STFT), while Nogav and Adeli^[201] processed the raw EEG with 1D-CNN. However, the data are usually not annotated in the real world; therefore, Weimann and Conrad^[16] evaluated the performance of the unsupervised pre-training model, and, as reported in the study, unsupervised or self-supervised pre-training yielded a lower performance than supervised pre-training, but they will become more relevant because they rely on fewer annotations.

To leverage the unlabelled bio-signals, like [16], Thin-

sungnoen et al.^[35] designed auto-encoder networks training bio-signal representations and then clustering features. In recent years, the emerging family of self-supervised (contrastive) learning methods has been applied to bio-signals data^[186-192, 199, 202-204]. Mehari and Strodtholt^[188] assessed self-supervised representation learning from 12-lead ECG data using SimCLR, BYOL, and CPC, from which CPC got the best results that only fell behind 0.5% supervised performance. Liu et al.^[189] also explored using the self-supervised learning approaches to detect arrhythmia; unlike [188], they converted ECG signals into grey-scale bitmap; additionally, they emphasised that the self-supervised learning can alleviate the problem of label imbalance and significantly reduce the quantity of requirement for annotated data. For EEG data, Mohsenvand et al.^[199] presented sequential contrastive learning of representation (SeqSLR) to diagnose epilepsy, which is based on the channel-wise feature extractor based on SimCLR, demonstrating that it outperforms conventional contrastive learning frameworks. A self-supervised pre-training framework, contrastive learning of cardiac signs (CLOCS), specifically designed for cardiac signals, is used to exploit the cardiac data across space, time, and patients^[186]. Zhang et al.^[191] proposed a general bio-signal framework referred to as time-frequency consistency (TF-C) by contrasting the samples in the time domain and the frequency domain, evaluating it in diagnosing the arrhythmia using ECG, epilepsy using EEG, and muscular diseases using EMG data. The latest work from Tang et al.^[202] proposed a self-supervised graph neural network to diagnose seizures on EEG, demonstrating that the self-supervised pre-training has consistently improved. It represents the spatial-temporal dependencies in EEGs using GNN and the self-supervised pretraining strategy to improve performance.

4.3 Emotion detection

Emotion detection has become an emerging field of study in computer-aided learning to equip machines with the ability to recognise the emotional states of individuals^[210]. An emotion can be considered a physiological and psychological expression that can be detected by many types of bio-signals, such as electrocardiograms (ECG), electrooculograms (EOG), galvanic skin responses (GSR), etc.^[195] The emotion computation analysis with DL and ML has achieved success, like stress or anxiety level detection^[194, 211], personality analysis^[212], emotion recognition^[193], etc. However, aside from the lack of data and annotations, emotion detection systems expect generalised models that can take into account the state of the emotion from a global perspective, and these generalised models can transfer to other tasks. Taking a multi-task generalised model, for example, can be transferred to a specific emotion task^[195], where the process of training this generalised model takes into account the pre-training techniques. To investigate if the pre-trained models enable enhanced performance in emotion detection, Radhika et al.^[193, 197, 198] pre-trained CNNs with annotated data and fine-tuning on target source data. Among them, Cimtay and Ekmekcioglu^[197] stated that they applied the pretrained model to cross-subject and cross-dataset EEG signals and reported promising results. In our survey, we have found that some recent emotion detection tasks learn the generalised pre-training feature representation through self-supervised learning methods. For instance, Sarkar and Etemad^[195] proposed a self-supervised network to pre-train the feature embeddings on an aggregation of four publicly available datasets to overcome the challenge of having different types of output labels for each dataset. In comparison to training on individual datasets, the framework with a pre-trained model performed better in emotion recognition. Furthermore, they proposed a self-supervised representation learning framework to detect maternal and fetal stress on ECG data and applied it in real-world practice^[194]. From EEG data, Mohsenvand et al.^[199] evaluated their proposed SecCLR framework in emotion recognition on the SEED dataset.

4.4 Sleep stage detection and other tasks

Sleep stage detection aims to determine the sleep stage from polysomnography (PSG), EEG, EOG, and EMG. Phan et al.^[207] proposed SeqSleepNet+ and DeepSleepNet+ frameworks with pre-trained models to classify sleep stages. They conducted pre-training on one type of signal and fine-tuning on another type. For example, they pre-trained the model on ECG and EOG source set and fine-tuned it on EEG and EOG target set. Pre-trained SeqSleepNet+ and DeepSleepNet+ models resulted in a significant improvement in sleep staging performance. Banville et al.^[204] investigated and explored using self-supervised learning to pre-train feature representations on EEG-based sleep staging detection. Jiang et al.^[208] proposed a self-supervised contrastive pre-training method to conduct representation learning of EEG signals applied to sleep stage tasks. With more unlabelled data available for the network, the proposed method reached 88.16% accuracy on the Sleep-EDF dataset. The model TF-C^[191] evaluated the performance on the sleep staging classification task.

Other bio-signal data tasks also employ the pre-training model approach to learn their feature representation. Aston et al.^[196] extracted features from the two-dimensional attractor generated from the ECG signal by the novel symmetric projection attractor reconstruction (SPAR) method used to detect a mouse's genetic mutation using pre-trained models that were trained on the ImageNet dataset. Identifying motor and mental imagery is a vital topic in brain-computer interface (BCI) research that recognises the subject's intention to such as implement prosthesis control^[213]. Amin et al.^[5] and Sadiq et al.^[205] utilised the fully-supervised pre-trained models to enhance the performance on the small EEG BCI datasets. Cheng et al.^[190], and Jiang et. al.^[208] proposed the self-supervised learning-based model to overcome challenge of performance degradation under a small number of labelled training samples.

4.5 Section conclusion

This section summarizes recent studies that pre-train feature representations and use the pre-trained model on downstream tasks on bio-signal data. Table 4 lists the related tasks and the corresponding citations. These studies have all shown the pre-training techniques to succeed in specific scenarios. However, many limitations exist in current studies. First, some studies have shown that pretraining does not lead to any notable improvements in the tasks. However, it can significantly reduce the training time and speed up the convergence process. Additionally, no large-scale dataset supports pre-training a generalized and high-quality representation of features. Therefore, for bio-signals, a specific pre-training framework is required to explore to get further improvements in the performance, such as CLOCS^[186], SeqSLR^[199], and TF-C^[191]. Furthermore, although Liu et al.^[189] showed that self-supervised learning can alleviate the class imbalance problem, the class imbalance has remained a standard issue for the bio-signal dataset, yet only a few works have attempted to address the issue.

5 EHRs in pre-training

In comparison with pre-training in other areas, there are fewer exploration opportunities for EHR data. In this section, we summarize the latest research for EHRs based on pre-training. As listed in Table 5, this table presents a compilation of various tasks related to Electronic Health Records (EHRs) along with relevant studies conducted in each of the tasks. Pre-training has been extremely successful in many areas. EHRs data-related tasks are one of

Identify motor mental imaginary

Sleep stage detection Multi-task the areas where pre-training has had a significant impact. In conditions where there is a lack of data, it is possible to enhance the performance of the model^[214]. The EHRs-related tasks include prediction^[33, 126, 214–222], information extraction from clinic notes^[223–226], the international classification of disease (ICD) coding^[227, 228], medication recommendation^[229, 230], etc.

5.1 Prediction

AI-aided prediction is critical in clinical practice, automatically analysing patients' conditions and providing suggestions to doctors for saving more lives. The primary purpose of prediction in EHRs is to predict the progression of the disease, such as mortality, the next visit, etc. For example, Rasmy et al.^[126] utilised the pre-trained model referred to as Med-BERT to predict heart failure among patients with diabetes and the onset of pancreatic cancer on the Truven and Cerner EHR datasets. They developed a domain-specific cross-visit pre-training model based on the BERT model. Med-BERT achieved promising performance on disease prediction tasks with small fine-tuning datasets and enabled to boost the AUC by more than 20%. However, if the pre-trained model consists of abundant auxiliary tasks and has a complex relationship to the target task, using the pre-trained model becomes inefficient and subnormal^[217]. Xue et al.^[217] proposed a method to automatically select from a large set of auxiliary tasks to address the challenge. They employed the self-supervised pre-training and the pre-trained model to predict clinical outcomes. Tipirneni and Reddy^[218] proposed a self-supervised transformer for the time-series model (STraTS) to predict clinical outcomes, which overcomes the challenges of sparsity and irregular time intervals in EHRs-related works; meanwhile, STraTS leverage unlabelled data for tackling the issue of limited availability of labelled data. McDermott et al.^[214] established a pre-training benchmark dataset for EHR time-series data to which various fine-tuning tasks are conducted, filling an essential hole and providing a baseline for pre-train-

[190, 206]

[191, 204, 208]

[209]

Datasets	Tasks	PT w labels	PT wt labels	Semi PT
ECG	De-noise	[180]	[181]	
	QRS detection			[182]
	Diagnosis/Classification	[15, 16, 107, 183 - 185]	[16, 35, 186 - 192]	
	Emotion detection	[193]	[194, 195]	
	Detection of genetic	[196]		
	Emotion detection	[197, 198]	[199]	
· · · · · · · · · · · · · · · · · · ·	Disease detection	[200, 201]	[191, 199, 202–204]	

[5, 205]

[207]

Table 4 Summary of bio-signal data in the medical domain based on pre-training. PT w labels: pre-training with labels; PT wt labels: pre-training without labels; Semi PT: pre-training using semi-supervised learning.

Deringer

EEG

Table 5 Summary of EHRs-related tasks based on pre-training

Tasks	Related papers		
Prediction	[33, 126, 214 – 222, 231, 232]		
Information extraction	[223–226]		
ICD coding	[227, 228]		
Medication recommendation	[229, 230]		

ing on EHR data. They evaluated the benchmarking with self-supervised pre-training and weakly-supervised multitask. Xu et al.^[216] introduced the medical knowledge graph combined with self-supervised pre-training to deal with the sparsity and high-dimensional issue of EHR data. Lu et al.^[221] utilised a pre-trained model to detect disease complications and compute the contributions of particular diseases and admissions. Using the self-supervised learning method, the pre-trained model was trained based on the hidden disease representation. Meng et al.^[33] proposed a model that can process five heterogeneous and high-dimensional datasets in a temporal manner in order to predict chronic diseases, such as depression. Aken et al.^[220] conducted clinical outcome pre-training to integrate knowledge about patient outcomes from multiple public sources. The model learns a representation for clinical outcomes, in which the model learns a relation between symptoms, risk factors, and clinical outcomes. Chen et al.^[215] proposed the physiological signal embeddings (PHASE) framework to forecast adverse surgical outcomes accurately. PHASE is a self-supervised-based model that learns the representations of the physiological signal and then uses the other prediction method to forecast the outcome. In addition, considering privacy issues, they attempted to simulate transferring the pre-trained model between organisations. The conventional sequential models are difficult to reuse for the early diagnosis of pregnancy complications; therefore, Ren et al.^[219] proposed the representation by pre-training time-aware transformer, particularly for the early diagnosis of pregnancy complications. In this task, they designed three pre-training tasks to handle data insufficiency, incompleteness, and short sequence problems. Hur et al.^[222] designed description-based embedding (December), a codeagnostic description-based representation learning framework for predictive tasks. They evaluated the performance of the proposed model on prediction tasks, transfer learning, and pooled learning. No uniform standard in EHRs is limited to applying the trained prediction models well to other EHR datasets from different organizations. To this end, Sun et al.^[231] proposed a generic transfer learning strategy that first pre-trains the model on source datasets then transfers the best-performing pretrained model to target datasets for fine-tuning the network. Ma et al.^[232] proposed a distil transfer learning framework, DistCare, for prognosis. DistCare leverages the existing EHR data, thus reducing the impact of the

available data limited to the rarity of cases and privacy issues. Specifically, they pre-trained models on the publicly available COVID-19-related EHRs, regarded as the teacher model based on distillation to obtain more comprehensive representations of source datasets. A series of extensive experiments on different clinical tasks and datasets show that DisCare benefits the prognosis with limited data.

5.2 Information extraction

EHR data contains valuable information which can assist doctors in diagnosing and making the treatment scheme. Recent studies have introduced pre-training in processing EHR data. For example, Chen et al.^[224] utilised the BERT pre-training EHR data embeddings to extract features with the structural data. The extracted features are then used to train with SimCLR and Deep InfoMax (DIM) under an unsupervised learning strategy to embed the disease concept. The pre-trained model further fine-tunes to adapt to the target outcome prediction task. Zhang et al.^[226] proposed the DeepEnroll model, which combines enrollment criteria and patient records into a shared latent space using a cross-modal inference learning approach. DeepEnroll encodes the patient's EHR data using the pre-trained BERT model. A real-world dataset was applied to this approach, and impressive results were achieved. Most existing studies do not consider capturing the time doctor experience and expertise with time-evolving in EHR data and learning static doctor representation. Biswal et al.^[223] proposed the Doctor2Vec model, which simultaneously enables learning the doctor representation and trial representation. The model achieved an 8.1% relative improvement in PR-AUC compared with the baseline model relying on dynamic doctor representation learning and pre-training a BERT model to understand trail descriptions. As the survey shows, some of the models on EHR data utilised the transformerbased model as the pre-train model, like BERT. However, in real-world clinical practice, there is an amount of privacy and sensitive information in the EHR data. In order to investigate whether the pre-trained embedding can be converted into the original information, thus causing privacy leakage, Lehman et al.^[225] executed experiments attempting to recover the personal healthcare information from the feature embeddings. They stated that a simple attack could not recover sensitive information, but more sophisticated methods could do this.

5.3 ICD classification

ICD coding is the task of predicting and coding all doctors' diagnoses with clinical test notes containing patients' symptoms and diagnostic procedures in an unstructured text format^[233]. Recent studies have provided evidence suggesting that DL and ML can classify the ICD coding. Data annotation is a time-consuming task, while for clinical text notes, annotation requires professional expertise. To solve the lack of annotation issue, some researchers focus on using pre-training methods^[227, 228]. Hlynsson et al.^[227] proposed a semi-self-supervised ICD coding framework. They attempted to train pre-trained models with four existing transformer-based models for clinical feature extraction and then use the data with the label to train a logistic regression ICD classifier. Zhang et al.^[228] utilised the BERT model to pre-train on a largescale dataset. Unlike Hlynsson's work, they introduced a multi-label attention method to train the classifier.

5.4 Medication recommendation

Medication recommendation is a hot topic in healthcare. It aims to recommend a set of medicines according to the patient's symptoms, which would play a critical role in assisting doctors in making decisions^[234]. Meanwhile, it could be a potential strategy to mitigate the doctor shortage problem in some countries. Technically, medication recommendation systems are trained on EHR data. Existing methods only utilise longitudinal EHRs with multiple visits while ignoring a large number of patients with a single visit. Shang et al.^[230] proposed the graph BERT (G-BERT) for medical code representation and medication recommendations to overcome this issue. G-BERT is the first model that introduced a language model pre-training strategy to the medical domain. Considering capturing local and global dependency information from records of patients, Su et al.^[229] proposed a dynamic time-aware hierarchical dependency network (TAHDNet) for medication recommendation tasks. The performance of the proposed method is superior to that of G-BERT.

5.5 Section conclusion

In this section, we summarised the recent advanced studies in pre-training on EHR data. These studies were based on four tasks: prediction, information extraction from EHRs, ICD coding, and medication recommendation. There is no doubt that the transformer-based model is the mainstream for EHR data pre-training-related works. Some recent studies utilised GNN as pre-training to improve performance and interpretability. The development of a privacy-related pre-training framework seems to be a promising topic in EHR studies, as discussed in [229].

6 Multi-modality in pre-training

Most publicly available healthcare datasets consist of multiple modalities. It is a part of nature, since the information that people are exposed to is always multimodal. People see the colour, hear the sound, feel the texture and smell the odour. Humans leverage different senses to better understand the information they receive. A significant reason to learn from multi-modal data is the assumption that the complementary nature of the different modalities can effectively improve performance. This assumption also applies to the medical domain. For instance, to write a comprehensive clinical report, clinicians need to review the patient's medical images and assess their medical history and vital signs. Cross-modal information could potentially improve the clinicians' understanding of patients' conditions.

Advances in uni-modal representation learning provide a firm foundation for improving performance in downstream tasks. The most common modalities in uni-modal pre-training are vision and language. In 2018, the advent of BERT^[128] significantly boosted representation learning in the area of NLP. Inspired by the success of uni-modal representation learning and BERT, researchers have started to look for methods to extract joint representations from multiple modalities. To date, most multi-modal pretrained models are based on visual and textual modalities. Li et al.^[235] adopted four transformer-based^[127] visionand-language (V+L) pre-trained models for medical downstream task, namely, VisualBERT^[236], Uniter^[237], Lxmert^[238], and PixelBERT^[239]. Moreover, they compared the performance of these models using AUC. According to the experimental results, Li et al.^[235] demonstrated that these four pre-trained V+L models outperformed the traditional CNN+RNN approach in the radiological classification task. Furthermore, Li et al.^[235] also showed the advantages of multi-modal pre-training over text-only embedding.

6.1 Multi-modal pre-training tasks in the medical domain

Almost all the multi-modal pre-training in the medical domain is based on V+L modalities. Therefore, most downstream tasks focus on medical images and clinical reports, such as radiology image interpretation and medical visual question answering (VQA). Radiological examination is one of the most common diagnostic procedures in medicine. Radiologists need to read a large number of radiology images daily. Introducing AI technology to generate diagnosis reports is crucial for radiology examinations, where a model is used to describe a medical image. Recent research has mainly focused on disease diagnosis and the generation of preliminary diagnosis reports. Most multi-modal studies in the medical field currently focus on this scenario. Generally, the multi-modality models in radiology are used for disease diagnosis and report generation. Wang et al.^[17] introduced an image-text pre-training approach that enables learning from raw data with mixed modality data, such as images and text. Most importantly, the data come from different institutions. In specific, the core structure of this method is a transformer-based self-supervised framework for simultaneously learning chest X-rays and corresponding text reports. They evaluated their model on three real-life application tasks: disease classification, similarity search and image regeneration. Wang et al.^[18] also proposed a large-scale chest X-ray dataset used to process images and corresponding reports, which provides a prominent expectation in this field. Similar to [17], Moon et al.^[19] explored the representation of learning tasks in the medical domain and proposed a transformer-based pre-training architecture with the multi-model attention masking scheme, namely medical vision language learner (Med-ViLL) for image-text understanding (e.g., diagnosis, medical image-text retrieval, and medical visual question answering) and version-language generation tasks (radiology report generation). In the extended experiments, MedViLL demonstrated the generalization ability under the transfer learning scenario on two chest X-ray datasets. Yan and Pei^[20] proposed a pre-training model, clinical-BERT, for three specific tasks, such as clinical diagnosis, masked medical subject headings (MeSH) words modelling and image-MeSH matching, through which the Clinical-BERT pre-train the model with the medical domain knowledge, rather than regarding the medical domain words and other words treated equally as MedViLL. They demonstrate that their proposed pre-training model is effective in downstream radiograph diagnosis and report generation tasks. Radiologists are always located in a small area with the most valuable information when they read medical images. In addition, many similar sentences describe generic image areas in the reports that are redundant and can be considered non-relevant noises^[156]. Most works ignored these issues. To address these issues and mimic radiology experts, Li et al.^[156] proposed an auxiliary signal-guided knowledge encoder-decoder (AS-GK) in which they pre-train a medical language model using the medical textual information they collected.

Since annotations of medical images require the participation of experts in corresponding domains, it is hard to obtain accurate labels of large-size datasets, and the cost is high. Therefore, only a small number of existing datasets can be used in the research of VQA on medical images. Thus, inspired by the success of self-supervised pretraining methods in NLP, vision, and language space, a multi-modal medical BERT (MMBERT)^[240] was proposed, which uses existing large multi-modal medical datasets to learn better image and text presentations. Compared to other state-of-the-art (SOTA) methods on VQA tasks on medical images, MMBERT achieves superior performance and provides attention maps to improve model interpretability.

7 Challenges and future directions

In Sections 3–6, we comprehensively reviewed and summarised the current state-of-the-art approaches using pre-training in the medical domain. For basic pre-training techniques, there are further development directions in the future study, such as improvement of computation efficiency both in the model pre-training and downstream tasks and the research for developing a none specific task models. Future research directions could be explored to maximise pre-training benefits in the medical domain. While many efforts have been devoted to this field, some challenges still need to be fully explored. In this section, we focus on discussing the challenges based on an analysis of the works mentioned above, which may stimulate a more profound study in the future. We outline several key research directions that were found when we summarised those works.

Data scarcity remains one of the most significant barriers to training a high-performance model for medical tasks. Although hospitals and other institutions can produce much healthcare data daily, those data cannot be available due to the increasingly strict data privacy clauses. Many datasets have been published for research, as we summarised in Section 2. However, the quantity of the data is still short for pre-training a general-purpose model, especially for bio-signal, EHR and multi-modalityrelated tasks. Therefore, it is a future direction to pretrain a general-purpose model on limited data.

Privacy concerns about healthcare data require urgent attention due to the strict data privacy clauses. Specifically, the question of whether personal healthcare information can be recovered via malicious attacks from the pre-trained feature representation has yet to be thoroughly investigated. This problem would influence whether or not pre-training techniques could be widely applied in real-world applications. Many privacy-related tasks in ML and DL have become hot topics, such as federated learning^[241] and differential privacy learning^[242]. It is expected that pre-training techniques combined with machine learning research relating to privacy will be a promising field for future research in the upcoming years. Many recent works have started to research this medical data privacy field, and this topic would be well worth studying.

Class imbalance is a common challenge in machine learning and deep learning. Especially in the medical domain, disease examples are always less than non-disease examples. For some rare diseases, the class imbalance issue will be extreme. If a deep learning model has been trained on a class-imbalanced dataset, the model will bias toward the majority category. Therefore, this problem is considered when we use a class-imbalanced dataset. However, we found only a few papers considering this problem in model pre-training after our investigation. Although most of the works train the model on an unlabelled dataset with unsupervised learning or self-supervised learning strategy, we could know a rough data distribution of the dataset. The imbalance issue will be considered in the training process.

In Section 6, we have introduced the multi-modality in pre-training in the medical domain. Many researchers have tried to introduce pre-training to process the multimodality data. However, most of the current research only focuses on generating clinical reports and tries to use the model to interpret the radiological examination, and the main reason is that there are many large datasets for this task. In contrast, the lack of task-related datasets limits the progress of research on multi-modality pretraining. Furthermore, there are currently few works on applying pre-training for bio-signal data to make survival predictions. Since bio-signal alone may not provide sufficient information for survival prediction, we see the potential of combining multi-modality data for this critical task and have included a discussion on this interesting future research direction.

8 Conclusions

Pre-training techniques are hot research topics in ML and DL. It has attracted much attention in medical domain due to the challenges posed by medical data, such as the data scarcity and lack of annotation. We review in detail the recent advances in pre-training-based frameworks for healthcare areas. This work proposes suggestions for physicians and researchers in AI who want to learn about the latest pre-training techniques in the medical domain. We briefly introduced the publicly available medical benchmark datasets and general pre-training strategies. Sections 3–6 investigate the extensive use of pre-training in different scenarios in the medical domain from four perspectives: images, bio-signal data, EHR data, and multi-modality data. At the end of this survey, we discuss the challenges and their possible solutions.

Acknowledgements

This work was supported by 2021 UQ School of Information Technology and Electrical Engineering (ITEE) Research Support Funding, Cyber Research Seed Funding (No. 2021-R3), the University of Adelaide (No. 1531570) and New Staff Research Start-up Funds (No. NS-2102). Open Access funding enabled and organized by CAUL and its Member Institutions.

Open Access

This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made.

The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder.

To view a copy of this licence, visit http://creative-commons.org/licenses/by/4.0/.

References

- Y. LeCun, B. Boser, J. S. Denker, D. Henderson, R. E. Howard, W. Hubbard, L. D. Jackel. Backpropagation applied to handwritten zip code recognition. *Neural Computation*, vol. 1, no. 4, pp. 541–551, 1989. DOI: 10.1162/neco. 1989.1.4.541.
- K. Simonyan, A. Zisserman. Very deep convolutional networks for large-scale image recognition. [Online], Available: https://arxiv.org/abs/1409.1556, 2014.
- [3] I. Sutskever, O. Vinyals, Q. V. Le. Sequence to sequence learning with neural networks. In *Proceedings of the 27th International Conference on Neural Information Processing Systems*, ACM, Montreal, Canada, pp.3104– 3112, 2014.
- [4] J. Chung, C. Gulcehre, K. Cho, Y. Bengio. Empirical evaluation of gated recurrent neural networks on sequence modeling. [Online], Available: https://arxiv.org/ abs/1412.3555, 2014.
- [5] S. U. Amin, M. Alsulaiman, G. Muhammad, M. A. Bencherif, M. S. Hossain. Multilevel weighted feature fusion using convolutional neural networks for EEG motor imagery classification. *IEEE Access*, vol.7, pp.18940– 18950, 2019. DOI: 10.1109/ACCESS.2019.2895688.
- [6] M. Jaderberg, K. Simonyan, A. Zisserman, K. Kavukcuoglu. Spatial transformer networks. In *Proceedings of the* 28th International Conference on Neural Information Processing Systems, ACM, Montreal, Canada, pp. 2017– 2025, 2015.
- [7] A. Krizhevsky, I. Sutskever, G. E. Hinton. ImageNet classification with deep convolutional neural networks. *Communications of the ACM*, vol. 60, no. 6, pp. 84–90, 2017. DOI: 10.1145/3065386.
- [8] J. Hirschberg, C. D. Manning. Advances in natural language processing. *Science*, vol. 349, no. 6245, pp. 261–266, 2015. DOI: 10.1126/science.aaa8685.
- [9] G. T. Wang, M. A. Zuluaga, W. Q. Li, R. Pratt, P. A. Patel, M. Aertsen, T. Doel, A. L. David, J. Deprest, S. Ourselin, T. Vercauteren. DeepIGeoS: A deep interactive geodesic framework for medical image segmentation. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, vol.41, no.7, pp.1559–1572, 2019. DOI: 10. 1109/TPAMI.2018.2840695.
- [10] S. Minaee, Y. Y. Boykov, F. Porikli, A. Plaza, N. Kehtarnavaz, D. Terzopoulos. Image segmentation using deep learning: A survey. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, vol.44, no.7, pp.3523– 3542, 2022. DOI: 10.1109/TPAMI.2021.3059968.
- [11] H. Greenspan, B. Van Ginneken, R. M. Summers. Guest editorial deep learning in medical imaging: Overview and future promise of an exciting new technique. *IEEE Transactions on Medical Imaging*, vol.35, no.5, pp.1153–1159, 2016. DOI: 10.1109/TMI.2016.2553401.
- [12] Y. D. Wang, W. T. Chen, D. C. Pi, L. Yue. Adversarially regularized medication recommendation model with multi-hop memory network. *Knowledge and Information Systems*, vol. 63, no. 1, pp. 125–142, 2021. DOI: 10.1007/ s10115-020-01513-9.
- [13] Y. D. Wang, W. T. Chen, D. C. Pi, L. Yue, S. Wang, M. Xu. Self-supervised adversarial distribution regularization for medication recommendation. In *Proceedings of the 30th International Joint Conference on Artificial In-*

telligence, Montreal, Canada, pp.3134–3140, 2021. DOI: 10.24963/ijcai.2021/431.

- [14] Y. X. Qiu, W. T. Chen, L. Yue, M. Xu, B. F. Zhu. STCT: Spatial-temporal conv-transformer network for cardiac arrhythmias recognition. In *Proceedings of the 17th International Conference on Advanced Data Mining and Applications*, Springer, Sydney, Australia, pp. 86–100, 2022. DOI: 10.1007/978-3-030-95405-5 7.
- [15] V. J. R. Ripoll, A. Wojdel, E. Romero, P. Ramos, J. Brugada. ECG assessment based on neural networks with pretraining. *Applied Soft Computing*, vol.49, pp.399– 406, 2016. DOI: 10.1016/j.asoc.2016.08.013.
- [16] K. Weimann, T. O. F. Conrad. Transfer learning for ECG classification. *Scientific Reports*, vol.11, no.1, Article number 5251, 2021. DOI: 10.1038/s41598-021-84374-8.
- [17] X. S. Wang, Z. Y. Xu, L. Tam, D. Yang, D. G. Xu. Selfsupervised image-text pre-training with mixed data in chest X-rays. [Online], Available: https://arxiv.org/abs/ 2103.16022, 2021.
- [18] X. S. Wang, Y. F. Peng, L. Lu, Z. Y. Lu, M. Bagheri, R. M. Summers. ChestX-ray8: Hospital-scale chest X-ray database and benchmarks on weakly-supervised classification and localization of common thorax diseases. In Proceedings of IEEE Conference on Computer Vision and Pattern Recognition, Honolulu, USA, pp.2097–2106, 2017. DOI: 10.1109/CVPR.2017.369.
- [19] J. H. Moon, H. Lee, W. Shin, Y. H. Kim, E. Choi. Multimodal understanding and generation for medical images and text via vision-language pre-training. *IEEE Journal* of Biomedical and Health Informatics, to be published. DOI: 10.1109/JBHI.2022.3207502.
- [20] B. Yan, M. T. Pei. Clinical-BERT: Vision-language pretraining for radiograph diagnosis and reports generation. In Proceedings of the AAAI Conference on Artificial Intelligence, vol.36, no.3, pp.2982–2990, 2022. DOI: 10. 1609/aaai.v36i3.20204.
- [21] L. Hou, D. Samaras, T. M. Kurc, Y. Gao, J. E. Davis, J. H. Saltz. Patch-based convolutional neural network for whole slide tissue image classification. In *Proceedings of IEEE Conference on Computer Vision and Pattern Recognition*, Las Vegas, USA, pp. 2424–2433, 2016. DOI: 10. 1109/CVPR.2016.266.
- [22] H. C. Shin, H. R. Roth, M. C. Gao, L. Lu, Z. Y. Xu, I. Nogues, J. H. Yao, D. Mollura, R. M. Summers. Deep convolutional neural networks for computer-aided detection: CNN architectures, dataset characteristics and transfer learning. *IEEE Transactions on Medical Imaging*, vol.35, no.5, pp.1285–1298, 2016. DOI: 10.1109/ TMI.2016.2528162.
- [23] T. Würfl, F. C. Ghesu, V. Christlein, A. Maier. Deep learning computed tomography. In Proceedings of the 19th International Conference on Medical Image Computing and Computer-assisted Intervention, Springer, Athens, Greece, pp. 432–440, 2016. DOI: 10.1007/978-3-319-46726-9_50.
- [24] B. Ramsundar, P. Eastman, P. Walters, V. Pande. Deep Learning for the Life Sciences: Applying Deep Learning to Genomics, Microscopy, Drug Discovery, and More, Sebastopol, USA: O'Reilly Media, 2019.
- [25] X. Han, Z. Y. Zhang, N. Ding, Y. X. Gu, X. Liu, Y. Q. Huo, J. Z. Qiu, Y. Yao, A. Zhang, L. Zhang, W. T. Han, M. L. Huang, Q. Jin, Y. Y. Lan, Y. Liu, Z. Y. Liu, Z. W. Lu, X. P. Qiu, R. H. Song, J. Tang, J. R. Wen, J. H.

Yuan, W. X. Zhao, J. Zhu. Pre-trained models: Past, present and future. *AI Open*, vol.2, pp.225–250, 2021. DOI: 10.1016/j.aiopen.2021.08.002.

- [26] L. Torrey, J. Shavlik. Transfer learning. Handbook of Research on Machine Learning Applications and Trends: Algorithms, Methods, and Techniques, E. S. Olivas, J. D. M. Guerrero, M. Martinez-Sober, J. R. Magdalena-Benedito, A. J. S. López, Eds., Hershey, USA: IGI Global, pp.242–264, 2010. DOI: 10.4018/978-1-60566-766-9. ch011.
- [27] S. J. Pan, Q. Yang. A survey on transfer learning. IEEE Transactions on Knowledge and Data Engineering, vol. 22, no. 10, pp. 1345–1359, 2010. DOI: 10.1109/TKDE. 2009.191.
- [28] V. Jain, E. Learned-Miller. Online domain adaptation of a pre-trained cascade of classifiers. In *Proceedings of IEEE Computer Vision and Pattern Recognition*, Colorado Springs, USA, pp.577–584, 2011. DOI: 10.1109/CV-PR.2011.5995317.
- [29] A. Newell, J. Deng. How useful is self-supervised pretraining for visual tasks? In Proceedings of IEEE/CVF Conference on Computer Vision and Pattern Recognition, IEEE, Seattle, USA, pp. 7345–7354, 2020. DOI: 10.1109/ CVPR42600.2020.00737.
- [30] Y. Z. Yang, Z. Xu. Rethinking the value of labels for improving class-imbalanced learning. In *Proceedings of the* 34th International Conference on Neural Information Processing Systems, ACM, Vancouver, Canada, pp. 19290– 19301, 2020.
- [31] H. Liu, J. Z. HaoChen, A. Gaidon, T. Y. Ma. Self-supervised learning is more robust to dataset imbalance. In Proceedings of the 10th International Conference on Learning Representations, 2022.
- [32] T. Schlegl, J. Ofner, G. Langs. Unsupervised pre-training across image domains improves lung tissue classification. In Proceedings of the International Workshop on Medical Computer Vision: Algorithms for Big Data, Springer, Cambridge, USA, pp.82–93, 2014. DOI: 10.1007/978-3-319-13972-2 8.
- [33] Y. W. Meng, W. Speier, M. K. Ong, C. W. Arnold. Bidirectional representation learning from transformers using multimodal electronic health record data to predict depression. *IEEE Journal of Biomedical and Health Informatics*, vol.25, no.8, pp.3121–3129, 2021. DOI: 10. 1109/JBHI.2021.3063721.
- [34] S. Azizi, B. Mustafa, F. Ryan, Z. Beaver, J. Freyberg, J. Deaton, A. Loh, A. Karthikesalingam, S. Kornblith, T. Chen, V. Natarajan, M. Norouzi. Big self-supervised models advance medical image classification. In *Proceedings of IEEE/CVF International Conference on Computer Vision*, IEEE, Montreal, Canada, pp.3478–3488, 2021. DOI: 10.1109/ICCV48922.2021.00346.
- [35] T. Thinsungnoen, K. Kerdprasop, N. Kerdprasop. Deep autoencoder networks optimized with genetic algorithms for efficient ECG clustering. *International Journal of Machine Learning and Computing*, vol.8, no.2, pp.112–116, 2018. DOI: 10.18178/ijmlc.2018.8.2.672.
- [36] V. Cheplygina, M. de Bruijne, J. P. W. Pluim. Not-so-supervised: A survey of semi-supervised, multi-instance, and transfer learning in medical image analysis. *Medical Image Analysis*, vol. 54, pp. 280–296, 2019. DOI: 10.1016/ j.media.2019.03.009.
- [37] T. D. Pham. A comprehensive study on classification of

COVID-19 on computed tomography with pretrained convolutional neural networks. *Scientific Reports*, vol. 10, no. 1, Article number 16942, 2020. DOI: 10.1038/s41598-020-74164-z.

- [38] P. Y. Chen. Representation learning for electronic health records: A survey. *Journal of Physics: Conference Series*, vol. 1487, Article number 012015, 2020. DOI: 10.1088/ 1742-6596/1487/1/012015.
- [39] S. Shurrab, R. Duwairi. Self-supervised learning methods and applications in medical imaging analysis: A survey. *PeerJ Computer Science*, vol.8, Article number e1045, 2022. DOI: 10.7717/peerj-cs.1045.
- [40] A. Ebbehoj, M.Ø. Thunbo, O. E. Andersen, M. V. Glindtvad, A. Hulman. Transfer learning for non-image data in clinical research: A scoping review. *PLoS Digital Health*, vol. 1, no. 2, Article number e0000014, 2022. DOI: 10.1371/journal.pdig.0000014.
- [41] T. J. Pollard, A. E. W. Johnson, J. D. Raffa, L. A. Celi, R. G. Mark, O. Badawi. The eICU collaborative research database, a freely available multi-center database for critical care research. *Scientific Data*, vol.5, no.1, Article number 180178, 2018. DOI: 10.1038/sdata.2018.178.
- [42] A. E. W. Johnson, T. J. Pollard, L. Shen, L. W. H. Lehman, M. L. Feng, M. Ghassemi, B. Moody, P. Szolovits, L. Anthony Celi, R. G. Mark. MIMIC-III, a freely accessible critical care database. *Scientific Data*, vol.3, no.1, Article number 160035, 2016. DOI: 10.1038/sdata.2016. 35.
- [43] A. L. Goldberger, L. A. Amaral, L. Glass, J. M. Hausdorff, P. C. Ivanov, R. G. Mark, J. E. Mietus, G. B. Moody, C. K. Peng, H. E. Stanley. PhysioBank, PhysioToolkit, and PhysioNet: Components of a new research resource for complex physiologic signals. *Circulation*, vol. 101, no. 23, pp. e215–e220, 2000. DOI: 10.1161/ 01.cir.101.23.e215.
- [44] A. E. Kavur, N. S. Gezer, M. Barış, S. Aslan, P. H. Conze, V. Groza, D. D. Pham, S. Chatterjee, P. Ernst, S. Özkan, B. Baydar, D. Lachinov, S. Han, J. Pauli, F. Isensee, M. Perkonigg, R. Sathish, R. Rajan, D. Sheet, G. Dovletov, O. Speck, A. Nürnberger, K. H. Maier-Hein, G. Bozdağı Akar, G. Ünal, O. Dicle, M. A. Selver. CHAOS challengecombined (CT-MR) healthy abdominal organ segmentation. *Medical Image Analysis*, vol.69, Article number 101950, 2021. DOI: 10.1016/j.media.2020.101950.
- [45] A. Sinha, J. Dolz. Multi-scale self-guided attention for medical image segmentation. *IEEE Journal of Biomedic*al and Health Informatics, vol.25, no.1, pp.121–130, 2021. DOI: 10.1109/JBHI.2020.2986926.
- [46] H. R. Roth, L. Lu, A. Farag, H. C. Shin, J. M. Liu, E. B. Turkbey, R. M. Summers. DeepOrgan: Multi-level deep convolutional networks for automated pancreas segmentation. In Proceedings of the 18th International Conference on Medical Image Computing and Computer-assisted Intervention, Springer, Munich, Germany, pp. 556–564, 2015. DOI: 10.1007/978-3-319-24553-9_68.
- [47] P. Bilic, P. F. Christ, E. Vorontsov, G. Chlebus, H. Chen, Q. Dou, C. W. Fu, X. Han, P. A. Heng, J. Hesser, S. Kadoury, T. K. Konopczynski, M. Le, C. M. Li, X. M. Li, J. Lipková, J. S. Lowengrub, H. Meine, J. H. Moltz, C. Pal, M. Piraud, X. J. Qi, J. Qi, M. Rempfler, K. Roth, A. Schenk, A. Sekuboyina, P. Zhou, C. Hülsemeyer, M. Beetz, F. Ettlinger, F. Grün, G. Kaissis, F. Lohöfer, R. Braren, J. Holch, F. Hofmann, W. H. Sommer, V. Heinemann, C. Jacobs, G. E. H. Mamani, B. van Ginneken, G.

Chartrand, A. Tang, M. Drozdzal, A. Ben-Cohen, E. Klang, M. M. Amitai, E. Konen, H. Greenspan, J. Moreau, A. Hostettler, L. Soler, R. Vivanti, A. Szeskin, N. Lev-Cohain, J. Sosna, L. Joskowicz, B. H. Menze. The liver tumor segmentation benchmark (LiTS). [Online], Available: https://arxiv.org/abs/1901.04056, 2019.

- [48] X. M. Li, H. Chen, X. J. Qi, Q. Dou, C. W. Fu, P. A. Heng. H-DenseUNet: Hybrid densely connected UNet for liver and tumor segmentation from CT volumes. *IEEE Transactions on Medical Imaging*, vol.37, no.12, pp. 2663–2674, 2018. DOI: 10.1109/TMI.2018.2845918.
- [49] D. S. Kermany, M. Goldbaum, W. J. Cai, C. C. S. Valentim, H. Y. Liang, S. L. Baxter, A. McKeown, G. Yang, X. K. Wu, F. B. Yan, J. Dong, M. K. Prasadha, J. Pei, M. Y. L. Ting, J. Zhu, C. Li, S. Hewett, J. Dong, I. Ziyar, A. Shi, R. Z. Zhang, L. H. Zheng, R. Hou, W. Shi, X. Fu, Y. O. Duan, V. A. N. Huu, C. Wen, E. D. Zhang, C. L. Zhang, O. L. Li, X. B. Wang, M. A. Singer, X. D. Sun, J. Xu, A. Tafreshi, M. A. Lewis, H. M. Xia, K. Zhang. Identifying medical diagnoses and treatable diseases by image-based deep learning. *Cell*, vol. 172, no.5, pp.1122–1131, 2018. DOI: 10.1016/j.cell.2018.02.010.
- [50] W. Al-Dhabyani, M. Gomaa, H. Khaled, A. Fahmy. Dataset of breast ultrasound images. *Data in Brief*, vol. 28, Article number 104863, 2020. DOI: 10.1016/j.dib. 2019.104863.
- [51] W. K. Moon, Y. W. Lee, H. H. Ke, S. H. Lee, C. S. Huang, R. F. Chang. Computer-aided diagnosis of breast ultrasound images using ensemble learning from convolutional neural networks. *Computer Methods and Programs in Biomedicine*, vol. 190, Article number 105361, 2020. DOI: 10.1016/j.cmpb.2020.105361.
- [52] P. Rajpurkar, J. Irvin, K. Zhu, B. Yang, H. Mehta, T. Duan, D. Ding, A. Bagul, C. Langlotz, K. Shpanskaya, M. P. Lungren, A. Y. Ng. CheXNet: Radiologist-level pneumonia detection on chest X-Rays with deep learning. [Online], Available: https://arxiv.org/abs/1711.05225, 2017.
- [53] Z. Wang, Y. X. Yin, J. P. Shi, W. Fang, H. S. Li, X. G. Wang. Zoom-in-Net: Deep mining lesions for diabetic retinopathy detection. In *Proceedings of the 20th International Conference on Medical Image Computing and Computer Assisted Intervention*, Springer, Quebec City, Canada, pp.267–275, 2017. DOI: 10.1007/978-3-319-66179-7_31.
- [54] N. C. F. Codella, D. Gutman, M. E. Celebi, B. Helba, M. A. Marchetti, S. W. Dusza, A. Kalloo, K. Liopyris, N. Mishra, H. Kittler, A. Halpern. Skin lesion analysis toward melanoma detection: A challenge at the 2017 international symposium on biomedical imaging (ISBI), hosted by the international skin imaging collaboration (IS-IC). In *Proceedings of 15th IEEE International Symposium on Biomedical Imaging*, Washington DC, USA, pp. 168– 172, 2018. DOI: 10.1109/ISBI.2018.8363547.
- [55] N. Gessert, M. Nielsen, M. Shaikh, R. Werner, A. Schlaefer. Skin lesion classification using ensembles of multi-resolution efficientNets with meta data. *MethodsX*, vol.7, Article number 100864, 2020. DOI: 10.1016/j.mex.2020. 100864.
- [56] C. Kandoth, M. D. McLellan, F. Vandin, K. Ye, B. F. Niu, C. Lu, M. C. Xie, Q. Y. Zhang, J. F. McMichael, M. A. Wyczałkowski, M. D. M. Leiserson, C. A. Miller, J. S. Welch, M. J. Walter, M. C. Wendl, T. J. Ley, R. K. Wilson, B. J. Raphael, L. Ding. Mutational landscape

and significance across 12 major cancer types. *Nature*, vol. 502, no. 7471, pp. 333–339, 2013. DOI: 10.1038/nature 12634.

- [57] J. W. Yao, X. L. Zhu, F. Y. Zhu, J. Z. Huang. Deep correlational learning for survival prediction from multimodality data. In *Proceedings of the 20th International Conference on Medical Image Computing and Computer Assisted Intervention*, Springer, Quebec City, Canada, pp. 406–414, 2017. DOI: 10.1007/978-3-319-66185-8 46.
- [58] P. Mobadersany, S. Yousefi, M. Amgad, D. A. Gutman, J. S. Barnholtz-Sloan, J. E. Velázquez Vega, D. J. Brat, L. A. D. Cooper. Predicting cancer outcomes from histology and genomics using convolutional networks. *Proceedings of the National Academy of Sciences of the United States of America*, vol.115, no.13, pp.E2970– E2979, 2018. DOI: 10.1073/pnas.1717139115.
- [59] National Lung Screening Trial Research Team. The national lung screening trial: Overview and study design. *Radiology*, vol. 258, no. 1, pp. 243–253, 2011. DOI: 10. 1148/radiol.10091808.
- [60] J. W. Yao, X. L. Zhu, J. Jonnagaddala, N. Hawkins, J. Z. Huang. Whole slide images based cancer survival prediction using attention guided deep multiple instance learning networks. *Medical Image Analysis*, vol.65, Article number 101789, 2020. DOI: 10.1016/j.media.2020.101 789.
- [61] B. S. Veeling, J. Linmans, J. Winkens, T. Cohen, M. Welling. Rotation equivariant CNNs for digital pathology. In Proceedings of the 21st International Conference on Medical Image Computing and Computer-assisted Intervention, Springer, Granada, Spain, pp.210–218, 2018. DOI: 10.1007/978-3-030-00934-2 24.
- [62] G. B. Moody, R. G. Mark. The impact of the MIT-BIH arrhythmia database. *IEEE Engineering in Medicine and Biology Magazine*, vol. 20, no. 3, pp. 45–50, 2001. DOI: 10. 1109/51.932724.
- [63] P. Wagner, N. Strodthoff, R. D. Bousseljot, D. Kreiseler, F. I. Lunze, W. Samek, T. Schaeffter. PTB-XL, a large publicly available electrocardiography dataset. *Scientific Data*, vol.7, no.1, Article number 154, 2020. DOI: 10. 1038/s41597-020-0495-6.
- [64] G. B. Moody, W. K. Muldrow, R. G. Mark. A noise stress test for arrhythmia detectors. *Computers in Cardiology*, vol. 11, no.3, pp. 381–384, 1984.
- [65] A. Taddei, G. Distante, M. Emdin, P. Pisani, G. B. Moody, C. Zeelenberg, C. Marchesi. The European ST-T database: Standard for evaluating systems for the analysis of ST-T changes in ambulatory electrocardiography. *European Heart Journal*, vol.13, no.9, pp.1164–1172, 1992. DOI: 10.1093/oxfordjournals.eurheartj.a060332.
- [66] G. D. Clifford, C. Y. Liu, B. Moody, L. W. H. Lehman, I. Silva, Q. Li, A. E. Johnson, R. G. Mark. AF classification from a short single lead ECG recording: The physioNet/ computing in cardiology challenge 2017. In *Proceedings of Computing in Cardiology*, IEEE, Rennes, France, 2017. DOI: 10.22489/CinC.2017.065-469.
- [67] F. Andreotti, O. Carr, M. A. F. Pimentel, A. Mahdi, M. De Vos. Comparing feature-based classifiers and convolutional neural networks to detect arrhythmia from short segments of ECG. In *Proceedings of Computing in Cardiology*, IEEE, Rennes, France, 2017. DOI: 10.22489/CinC. 2017.360-239.
- [68] R. Bousseljot, D. Kreiseler, A. Schnabel. Nutzung der

EKG-signaldatenbank cardiodat der PTB über das internet. *Biomedizinische Technik*, vol. 40, Article number 317, 1995. DOI: 10.1515/bmte.1995.40.s1.317.

- [69] L. Sharma, R. Tripathy, S. Dandapat. Multiscale energy and eigenspace approach to detection and localization of myocardial infarction. *IEEE Transactions on Biomedical Engineering*, vol. 62, no. 7, pp. 1827–1837, 2015.
- [70] G. B. Moody, R. G. Mark. Development and evaluation of a 2-lead ECG analysis program. *Computers in Cardiology*, vol. 1982, pp. 39–44, 1982.
- [71] F. F. Liu, C. Y. Liu, L. N. Zhao, X. Y. Zhang, X. L. Wu, X. Y. Xu, Y. L. Liu, C. Y. Ma, S. S. Wei, Z. Q. He, J. Q. Li, E. N. Yin Kwee. An open access database for evaluating the algorithms of electrocardiogram rhythm and morphology abnormality detection. *Journal of Medical Imaging and Health Informatics*, vol.8, no.7, pp. 1368–1373, 2018. DOI: 10.1166/jmihi.2018.2442.
- [72] T. M. Chen, C. H. Huang, E. S. C. Shih, Y. F. Hu, M. J. Hwang. Detection and classification of cardiac arrhythmias by a challenge-best deep learning neural network model. *iScience*, vol.23, no.3, Article number 100886, 2020. DOI: 10.1016/j.isci.2020.100886.
- [73] J. A. Miranda-Correa, M. K. Abadi, N. Sebe, I. Patras. AMIGOS: A dataset for affect, personality and mood research on individuals and groups. *IEEE Transactions on Affective Computing*, vol.12, no.2, pp.479–493, 2021. DOI: 10.1109/TAFFC.2018.2884461.
- [74] L. Santamaria-Granados, M. Munoz-Organero, G. Ramirez-González, E. Abdulhay, N. Arunkumar. Using deep convolutional neural network for emotion detection on a physiological signals dataset (AMIGOS). *IEEE Access*, vol. 7, pp. 57–67, 2018. DOI: 10.1109/ACCESS.2018. 2883213.
- [75] R. Subramanian, J. Wache, M. K. Abadi, R. L. Vieriu, S. Winkler, N. Sebe. ASCERTAIN: Emotion and personality recognition using commercial sensors. *IEEE Transactions on Affective Computing*, vol.9, no.2, pp.147–160, 2018. DOI: 10.1109/TAFFC.2016.2625250.
- [76] L. Zhang, S. Walter, X. Y. Ma, P. Werner, A. Al-Hamadi, H. C. Traue, S. Gruss. "BioVid Emo DB": A multimodal database for emotion analyses validated by subjective ratings. In *Proceedings of IEEE Symposium Series on Computational Intelligence*, Athens, Greece, 2016. DOI: 10.1109/SSCI.2016.7849931.
- [77] Z. Cheng, L. Shu, J. Y. Xie, C. L. P. Chen. A novel ECGbased real-time detection method of negative emotions in wearable applications. In *Proceedings of International Conference on Security, Pattern Analysis, and Cybernetics*, IEEE, Shenzhen, China, pp. 296–301, 2017. DOI: 10. 1109/SPAC.2017.8304293.
- [78] S. Koelstra, C. Muhl, M. Soleymani, J. S. Lee, A. Yazdani, T. Ebrahimi, T. Pun, A. Nijholt, I. Patras. DEAP: A database for emotion analysis; using physiological signals. *IEEE Transactions on Affective Computing*, vol.3, no.1, pp.18–31, 2012. DOI: 10.1109/T-AFFC. 2011.15.
- [79] Z. Yin, M. Y. Zhao, Y. X. Wang, J. D. Yang, J. H. Zhang. Recognition of emotions using multimodal physiological signals and an ensemble deep learning model. *Computer Methods and Programs in Biomedicine*, vol.140, pp. 93–110, 2017. DOI: 10.1016/j.cmpb.2016.12.005.
- [80] S. Katsigiannis, N. Ramzan. DREAMER: A database for emotion recognition through EEG and ECG signals from

wireless low-cost off-the-shelf devices. *IEEE Journal of Biomedical and Health Informatics*, vol.22, no.1, pp.98–107, 2018. DOI: 10.1109/JBHI.2017.2688239.

- [81] T. F. Song, W. M. Zheng, P. Song, Z. Cui. EEG emotion recognition using dynamical graph convolutional neural networks. *IEEE Transactions on Affective Computing*, vol. 11, no. 3, pp. 532–541, 2020. DOI: 10.1109/TAFFC. 2018.2817622.
- [82] M. Soleymani, J. Lichtenauer, T. Pun, M. Pantic. A multimodal database for affect recognition and implicit tagging. *IEEE Transactions on Affective Computing*, vol.3, no.1, pp. 42–55, 2012. DOI: 10.1109/T-AFFC.2011.25.
- [83] X. B. Li, J. Chen, G. Y. Zhao, M. Pietikäinen. Remote heart rate measurement from face videos under realistic situations. In *Proceedings of IEEE Conference on Computer Vision and Pattern Recognition*, Columbus, USA, pp. 4264–4271, 2014. DOI: 10.1109/CVPR.2014.543.
- [84] T. F. Song, W. M. Zheng, C. Lu, Y. Zong, X. L. Zhang, Z. Cui. MPED: A multi-modal physiological emotion database for discrete emotion recognition. *IEEE Access*, vol. 7, pp.12177–12191, 2019. DOI: 10.1109/ACCESS.2019. 2891579.
- [85] W. L. Zheng, B. L. Lu. Investigating critical frequency bands and channels for EEG-based emotion recognition with deep neural networks. *IEEE Transactions on Autonomous Mental Development*, vol.7, no.3, pp.162– 175, 2015. DOI: 10.1109/TAMD.2015.2431497.
- [86] I. Obeid, J. Picone. The temple university hospital EEG data corpus. *Frontiers in Neuroscience*, vol. 10, Article number 196, 2016. DOI: 10.3389/fnins.2016.00196.
- [87] G. Schalk, D. J. McFarland, T. Hinterberger, N. Birbaumer, J. R. Wolpaw. BCI2000: A general-purpose brain-computer interface (BCI) system. *IEEE Transactions on Biomedical Engineering*, vol. 51, no. 6, pp.1034– 1043, 2004. DOI: 10.1109/TBME.2004.827072.
- [88] A. Supratak, H. Dong, C. Wu, Y. K. Guo. DeepSleepNet: A model for automatic sleep stage scoring based on raw single-channel EEG. *IEEE Transactions on Neural Systems and Rehabilitation Engineering*, vol.25, no.11, pp.1998–2008, 2017. DOI: 10.1109/TNSRE.2017.2721 116.
- [89] G. Q. Zhang, L. C. Cui, R. Mueller, S. Q. Tao, M. Kim, M. Rueschman, S. Mariani, D. Mobley, S. Redline. The national sleep research resource: Towards a sleep data commons. *Journal of the American Medical Informatics Association*, vol.25, no.10, pp.1351–1358, 2018. DOI: 10. 1093/jamia/ocy064.
- [90] S. F. Quan, B. V. Howard, C. Iber, J. P. Kiley, F. J. Nieto, G. T. O'Connor, D. M. Rapoport, S. Redline, J. Robbins, J. M. Samet, P. W. Wahl. The sleep heart health study: Design, rationale, and methods. *Sleep*, vol.20, no.12, pp.1077–1085, 1997.
- [91] A. Sors, S. Bonnet, S. Mirek, L. Vercueil, J. F. Payen. A convolutional neural network for sleep stage scoring from raw single-channel EEG. *Biomedical Signal Processing* and Control, vol.42, pp.107–114, 2018. DOI: 10.1016/j. bspc.2017.12.001.
- [92] Y. Bengio, P. Lamblin, D. Popovici, H. Larochelle. Greedy layer-wise training of deep networks. In Proceedings of the 19th International Conference on Neural Information Processing Systems, ACM, Vancouver, Canada, pp. 153–160, 2006.

- [93] M. Ranzato, Y. L. Boureau, Y. LeCun. Sparse feature learning for deep belief networks. In Proceedings of the 20th International Conference on Neural Information Processing Systems, ACM, Vancouver, Canada, pp. 1185– 1192, 2007.
- [94] K. M. He, R. Girshick, P. Dollár. Rethinking ImageNet pre-training. In Proceedings of IEEE/CVF International Conference on Computer Vision, IEEE, Seoul, Republic of Korea, pp. 4918–4927, 2019. DOI: 10.1109/ICCV.2019. 00502.
- [95] M. Raghu, C. Y. Zhang, J. Kleinberg, S. Bengio. Transfusion: Understanding transfer learning for medical imaging. In *Proceedings of the 33rd International Conference* on Neural Information Processing Systems, ACM, Vancouver, Canada, pp. 3347–3357, 2019.
- [96] S. Thrun, L. Pratt. Learning to learn: Introduction and overview. *Learning to Learn*, S. Thrun, L. Pratt, Eds., Boston, USA: Springer, pp.3–17, 1998. DOI: 10.1007/ 978-1-4615-5529-2_1.
- [97] H. Scudder. Probability of error of some adaptive pattern-recognition machines. *IEEE Transactions on Information Theory*, vol. 11, no.3, pp. 363–371, 1965. DOI: 10.1109/TIT.1965.1053799.
- [98] K. M. He, X. Y. Zhang, S. Q. Ren, J. Sun. Deep residual learning for image recognition. In *Proceedings of IEEE Conference on Computer Vision and Pattern Recognition*, Vegas, USA, pp. 770–778, 2016. DOI: 10.1109/CV-PR.2016.90.
- [99] C. Szegedy, W. Liu, Y. Q. Jia, P. Sermanet, S. Reed, D. Anguelov, D. Erhan, V. Vanhoucke, A. Rabinovich. Going deeper with convolutions. In *Proceedings of IEEE Conference on Computer Vision and Pattern Recognition*, Boston, USA, 2015. DOI: 10.1109/CVPR.2015. 7298594.
- [100] G. Huang, Z. Liu, L. Van Der Maaten, K. Q. Weinberger. Densely connected convolutional networks. In Proceedings of IEEE Conference on Computer Vision and Pattern Recognition, Honolulu, USA, pp. 2261–2269, 2017. DOI: 10.1109/CVPR.2017.243.
- [101] J. Sarzynska-Wawer, A. Wawer, A. Pawlak, J. Szymanowska, I. Stefaniak, M. Jarkiewicz, L. Okruszek. Detecting formal thought disorder by deep contextualized word representations. *Psychiatry Research*, vol. 304, Article number 114135, 2021. DOI: 10.1016/j.psychres.2021. 114135.
- [102] Z. Y. Han, B. Z. Wei, Y. J. Zheng, Y. L. Yin, K. J. Li, S. Li. Breast cancer multi-classification from histopathological images with structured deep learning model. *Scientific Reports*, vol.7, no.1, Article number 4172, 2017. DOI: 10.1038/s41598-017-04075-z.
- [103] A. Esteva, B. Kuprel, R. A. Novoa, J. Ko, S. M. Swetter, H. M. Blau, S. Thrun. Dermatologist-level classification of skin cancer with deep neural networks. *Nature*, vol.542, no.7639, pp.115–118, 2017. DOI: 10.1038/ nature21056.
- [104] J. De Fauw, J. R. Ledsam, B. Romera-Paredes, S. Nikolov, N. Tomasev, S. Blackwell, H. Askham, X. Glorot, B. O'Donoghue, D. Visentin, G. Van Den Driessche, B. Lakshminarayanan, C. Meyer, F. Mackinder, S. Bouton, K. Ayoub, R. Chopra, D. King, A. Karthikesalingam, C. O. Hughes, R. Raine, J. Hughes, D. A. Sim, C. Egan, A. Tufail, H. Montgomery, D. Hassabis, G. Rees, T. Back, P. T. Khaw, M. Suleyman, J. Cornebise, P. A. Keane, O. Ron-

neberger. Clinically applicable deep learning for diagnosis and referral in retinal disease. *Nature Medicine*, vol. 24, no.9, pp.1342–1350, 2018. DOI: 10.1038/s41591-018-0107-6.

- [105] M. Treder, J. L. Lauermann, N. Eter. Automated detection of exudative age-related macular degeneration in spectral domain optical coherence tomography using deep learning. Graefe's Archive for Clinical and Experimental Ophthalmology, vol.256, no.2, pp.259–265, 2018. DOI: 10.1007/s00417-017-3850-3.
- [106] I. D. Apostolopoulos, T. A. Mpesiana. Covid-19: Automatic detection from X-ray images utilizing transfer learning with convolutional neural networks. *Physical* and Engineering Sciences in Medicine, vol.43, no.2, pp. 635–640, 2020. DOI: 10.1007/s13246-020-00865-4.
- [107] M. M. Al Rahhal, Y. Bazi, M. Al Zuair, E. Othman, B. BenJdira. Convolutional neural networks for electrocardiogram classification. *Journal of Medical and Biological Engineering*, vol. 38, no. 6, pp. 1014–1025, 2018. DOI: 10. 1007/s40846-018-0389-7.
- [108] F. Demir, A. Sengur, V. Bajaj. Convolutional neural networks based efficient approach for classification of lung diseases. *Health Information Science and Systems*, vol.8, no.1, Article number 4, 2020. DOI: 10.1007/s13755-019-0091-3.
- [109] H. T. Shi, H. R. Wang, C. J. Qin, L. Q. Zhao, C. L. Liu. An incremental learning system for atrial fibrillation detection based on transfer learning and active learning. *Computer Methods and Programs in Biomedicine*, vol. 187, Article number 105219, 2020. DOI: 10.1016/j. cmpb.2019.105219.
- [110] A. Shyam, V. Ravichandran, S. P. Preejith, J. Joseph, M. Sivaprakasam. PPGnet: Deep network for device independent heart rate estimation from photoplethysmogram. In Proceedings of the 41st Annual International Conference of the IEEE Engineering in Medicine and Biology Society, IEEE, Berlin, Germany, pp.1899–1902, 2019. DOI: 10.1109/EMBC.2019.8856989.
- [111] Y. K. Li, S. Rao, J. R. A. Solares, A. Hassaine, R. Ramakrishnan, D. Canoy, Y. J. Zhu, K. Rahimi, G. Salimi-Khorshidi. BEHRT: Transformer for electronic health records. *Scientific Reports*, vol. 10, no. 1, Article number 7155, 2020. DOI: 10.1038/s41598-020-62922-y.
- [112] C. Matsoukas, J. F. Haslum, M. Sorkhei, M. Söderberg, K. Smith. What makes transfer learning work for medical images: Feature reuse & other factors. In *Proceedings* of *IEEE/CVF Conference on Computer Vision and Pattern Recognition*, IEEE, New Orleans, USA, pp.9215– 9224, 2022. DOI: 10.1109/CVPR52688.2022.00901.
- [113] T. Chen, S. Kornblith, M. Norouzi, G. Hinton. A simple framework for contrastive learning of visual representations. In *Proceedings of the 37th International Conference on Machine Learning*, pp. 1597–1607, 2020.
- [114] J. B. Grill, F. Strub, F. Altché, C. Tallec, P. H. Richemond, E. Buchatskaya, C. Doersch, B. Avila Pires, Z. D. Guo, M. Gheshlaghi Azar, B. Piot, K. Kavukcuoglu, R. Munos, M. Valko. Bootstrap your own latent a new approach to self-supervised learning. In *Proceedings of the 34th International Conference on Neural Information Processing Systems*, ACM, Vancouver, Canada, pp. 21271– 21284, 2020.
- [115] M. Ishan, L. V. D. Maaten. Self-supervised learning of pretext-invariant representations. In *Proceedings of the*

IEEE/CVF Conference on Computer Vision and Pattern Recognition, pp.6707–6717. 2020. DOI: 10.1109/cvpr42 600.2020.00674.

- [116] M. Caron, P. Bojanowski, J. Mairal, A. Joulin. Unsupervised pre-training of image features on non-curated data. In Proceedings of IEEE/CVF International Conference on Computer Vision, IEEE, Seoul, Republic of Korea, pp. 2959–2968, 2019. DOI: 10.1109/ICCV.2019.00305.
- [117] T. Chen, S. Kornblith, K. Swersky, M. Norouzi, G. E. Hinton. Big self-supervised models are strong semi-supervised learners. In *Proceedings of the 34th Conference on Neural Information Processing Systems*, Vancouver, Canada, pp. 22243–22255, 2020.
- [118] A. Van Den Oord, Y. Z. Li, O. Vinyals. Representation learning with contrastive predictive coding. [Online], Available: https://arxiv.org/abs/1807.03748, 2018.
- [119] K. M. He, H. Q. Fan, Y. X. Wu, S. N. Xie, R. Girshick. Momentum contrast for unsupervised visual representation learning. In *Proceedings of IEEE/CVF Conference* on Computer Vision and Pattern Recognition, IEEE, Seattle, USA, pp.9729–9738, 2020. DOI: 10.1109/ CVPR42600.2020.00975.
- [120] M. Caron, I. Misra, J. Mairal, P. Goyal, P. Bojanowski, A. Joulin. Unsupervised learning of visual features by contrasting cluster assignments. In *Proceedings of the* 34th International Conference on Neural Information Processing Systems, ACM, Vancouver, Canada, pp. 9912–9924, 2020.
- [121] K. M. He, X. L. Chen, S. N. Xie, Y. H. Li, P. Dollár, R. Girshick. Masked autoencoders are scalable vision learners. In *Proceedings of IEEE/CVF Conference on Computer Vision and Pattern Recognition*, IEEE, New Orleans, USA, pp. 15979–15988, 2022. DOI: 10.1109/CV-PR52688.2022.01553.
- [122] X. L. Chen, K. M. He. Exploring simple Siamese representation learning. In *Proceedings of IEEE/CVF Conference on Computer Vision and Pattern Recognition*, IEEE, Nashville, USA, pp.15745–15753, 2021. DOI: 10. 1109/CVPR46437.2021.01549.
- [123] O. J. Hénaff. Data-efficient image recognition with contrastive predictive coding. In *Proceedings of the 37th International Conference on Machine Learning*, pp. 4182– 4192, 2020.
- [124] X. L. Chen, H. Q. Fan, R. Girshick, K. M. He. Improved baselines with momentum contrastive learning. [Online], Available: https://arxiv.org/abs/2003.04297, 2020.
- [125] X. L. Chen, S. N. Xie, K. M. He. An empirical study of training self-supervised vision transformers. In Proceedings of IEEE/CVF International Conference on Computer Vision, IEEE, Montreal, Canada, pp.9620–9629, 2021. DOI: 10.1109/ICCV48922.2021.00950.
- [126] L. Rasmy, Y. Xiang, Z. Q. Xie, C. Tao, D. G. Zhi. Med-BERT: Pretrained contextualized embeddings on largescale structured electronic health records for disease prediction. NPJ Digital Medicine, vol. 4, no. 1, Article number 86, 2021. DOI: 10.1038/s41746-021-00455-y.
- [127] A. Vaswani, N. Shazeer, N. Parmar, J. Uszkoreit, L. Jones, A. N. Gomez, Ł. Kaiser, I. Polosukhin. Attention is all you need. In *Proceedings of the 31st International Conference on Neural Information Processing Systems*, ACM, Long Beach, USA, pp. 6000–6010, 2017.
- [128] J. Devlin, M. W. Chang, K. Lee, K. Toutanova. BERT:

Pre-training of deep bidirectional transformers for language understanding. In *Proceedings of Conference of the North American Chapter of the Association for Computational Linguistics: Human Language Technologies*, Minneapolis, USA, pp. 4171–4186, 2019. DOI: 10.18653/v1/ N19-1423.

- [129] S. M. Anwar, M. Majid, A. Qayyum, M. Awais, M. Alnowami, M. K. Khan. Medical image analysis using convolutional neural networks: A review. *Journal of Medical Systems*, vol.42, no.11, Article number 226, 2018. DOI: 10.1007/s10916-018-1088-1.
- [130] R. Paul, S. H. Hawkins, L. O. Hall, D. B. Goldgof, R. J. Gillies. Combining deep neural network and traditional image features to improve survival prediction accuracy for lung cancer patients from diagnostic CT. In *Proceedings of IEEE International Conference on Systems, Man, and Cybernetics*, Budapest, Hungary, pp.2570–2575, 2016. DOI: 10.1109/SMC.2016.7844626.
- [131] M. W. Ren, N. Dey, M. A. Styner, K. Botteron, G. Gerig. Local spatiotemporal representation learning for longitudinally-consistent neuroimage analysis. [Online], Available: https://arxiv.org/abs/2206.04281, 2022.
- [132] A. Bhandary, G. A. Prabhu, V. Rajinikanth, K. P. Thanaraj, S. C. Satapathy, D. E. Robbins, C. Shasky, Y. D. Zhang, J. M. R. Tavares, N. S. M. Raja. Deep-learning framework to detect lung abnormality-A study with chest X-ray and lung CT scan images. *Pattern Recognition Letters*, vol. 129, pp. 271–278, 2020. DOI: 10.1016/j. patrec.2019.11.013.
- [133] D. S. Reddy, R. Bharath, P. Rajalakshmi. A novel computer-aided diagnosis framework using deep learning for classification of fatty liver disease in ultrasound imaging. In Proceedings of the 20th IEEE International Conference on E-health Networking, Applications and Services (Healthcom), Ostrava, Czech Republic, 2018. DOI: 10. 1109/HealthCom.2018.8531118.
- [134] C. Z. Wu, J. Sun, J. Wang, L. F. Xu, S. Zhan. Encodingdecoding network with pyramid self-attention module for retinal vessel segmentation. *International Journal of Automation and Computing*, vol.18, no.6, pp.973–980, 2021. DOI: 10.1007/s11633-020-1277-0.
- [135] J. Ker, L. P. Wang, J. Rao, T. Lim. Deep learning applications in medical image analysis. *IEEE Access*, vol.6, pp. 9375–9389, 2017. DOI: 10.1109/ACCESS.2017.2788 044.
- [136] A. Fernandez-Quilez. Deep Learning for an Improved Diagnostic Pathway of Prostate Cancer in a Small Multi-Parametric Magnetic Resonance Data Regime, Ph.D. dissertation, University of Stavanger, Stavanger, Norway, 2022.
- [137] K. B. Ahmed, L. O. Hall, D. B. Goldgof, R. H. Liu, R. A. Gatenby. Fine-tuning convolutional deep features for MRI based brain tumor classification. In *Proceedings of SPIE 10134*, *Medical Imaging 2017: Computer-Aided Diagnosis*, Orlando, USA, Article number 101342E, 2017. DOI: 10.1117/12.2253982.
- [138] R. M. Prakash, R. S. S. Kumari. Classification of MR brain images for detection of tumor with transfer learning from pre-trained CNN models. In *Proceedings of the* 2019 International Conference on Wireless Communications Signal Processing and Networking, IEEE, Chennai, India, pp. 508–511, 2019. DOI: 10.1109/WiSPNET45539. 2019.9032811.

- [139] H. A. Khan, W. Jue, M. Mushtaq, M. U. Mushtaq. Brain tumor classification in MRI image using convolutional neural network. *Mathematical Biosciences and Engineering*, vol.17, no.5, pp.6203–6216, 2020. DOI: 10.3934/ mbe.2020328.
- [140] M. Sajjad, S. Khan, K. Muhammad, W. Q. Wu, A. Ullah, S. W. Baik. Multi-grade brain tumor classification using deep CNN with extensive data augmentation. *Journal of Computational Science*, vol. 30, pp. 174–182, 2019. DOI: 10.1016/j.jocs.2018.12.003.
- [141] S. Deepak, P. M. Ameer. Brain tumor classification using deep CNN features via transfer learning. *Computers in Biology and Medicine*, vol. 111, Article number 103345, 2019. DOI: 10.1016/j.compbiomed.2019.103345.
- [142] N. Noreen, S. Palaniappan, A. Qayyum, I. Ahmad, M. Imran, M. Shoaib. A deep learning model based on concatenation approach for the diagnosis of brain tumor. *IEEE Access*, vol.8, pp.55135–55144, 2020. DOI: 10. 1109/ACCESS.2020.2978629.
- [143] J. Cheng. Brain tumor dataset. Figshare, [Online], Available: https://doi.org/10.6084/m9.figshare.1512427. v5, 2017.
- [144] Z. N. K. Swati, Q. H. Zhao, M. Kabir, F. Ali, Z. Ali, S. Ahmed, J. F. Lu. Brain tumor classification for MR images using transfer learning and fine-tuning. *Computerized Medical Imaging and Graphics*, vol.75, pp.34–46, 2019. DOI: 10.1016/j.compmedimag.2019.05.001.
- [145] F. J.Díaz-Pernas, M. Martínez-Zarzuela, M. Antón-Rodríguez, D.González-Ortega. A deep learning approach for brain tumor classification and segmentation using a multiscale convolutional neural network. *Healthcare*, vol.9, no.2, Article number 153, 2021. DOI: 10. 3390/healthcare9020153.
- [146] S. D. Wang, L. Y. Dong, X. Wang, X. G. Wang. Classification of pathological types of lung cancer from CT images by deep residual neural networks with transfer learning strategy. *Open Medicine*, vol. 15, no. 1, pp. 190–197, 2020. DOI: 10.1515/med-2020-0028.
- [147] P. Marentakis, P. Karaiskos, V. Kouloulias, N. Kelekis, S. Argentos, N. Oikonomopoulos, C. Loukas. Lung cancer histology classification from CT images based on radiomics and deep learning models. *Medical & Biological Engineering & Computing*, vol.59, no.1, pp.215–226, 2021. DOI: 10.1007/s11517-020-02302-w.
- [148] H. Kutlu, E. Avcı. A novel method for classifying liver and brain tumors using convolutional neural networks, discrete wavelet transform and long short-term memory networks. Sensors, vol.19, no.9, Article number 1992, 2019. DOI: 10.3390/s19091992.
- [149] M. Byra, G. Styczynski, C. Szmigielski, P. Kalinowski, Ł. Michałowski, R. Paluszkiewicz, B. Ziarkiewicz-Wróblewska, K. Zieniewicz, P. Sobieraj, A. Nowicki. Transfer learning with deep convolutional neural network for liver steatosis assessment in ultrasound images. International Journal of Computer Assisted Radiology and Surgery, vol. 13, no. 12, pp. 1895–1903, 2018. DOI: 10. 1007/s11548-018-1843-2.
- [150] M. Alkhaleefah, S. C. Ma, Y. L. Chang, B. Huang, P. K. Chittem, V. P. Achhannagari. Double-shot transfer learning for breast cancer classification from X-ray images. *Applied Sciences*, vol. 10, no. 11, Article number 3999, 2020. DOI: 10.3390/app10113999.
- [151] S. G. Armato III, G. McLennan, L. Bidaut, M. F. McNitt-

Gray, C. R. Meyer, A. P. Reeves, B. Zhao, D. R. Aberle, C. I. Henschke, E. A. Hoffman, E. A. Kazerooni, H. Macmahon, E. J. R. Van Beek, D. Yankelevitz, A. M. Biancardi, P. H. Bland, M. S. Brown, R. M. Engelmann, G. E. Laderach, D. Max, R. C. Pais, D. P. Y. Qing, R. Y. Roberts, A. R. Smith, A. Starkey, P. Batra, P. Caligiuri, A. Farooqi, G. W. Gladish, C. M. Jude, R. F. Munden, I. Petkovska, L. E. Quint, L. H. Schwartz, B. Sundaram, L. E. Dodd, C. Fenimore, D. Gur, N. Petrick, J. Freymann, J. Kirby, B. Hughes, A. Vande Casteele, S. Gupte, M. Sallam, M. D. Heath, M. H. Kuhn, E. Dharaiya, R. Burns, D. S. Fryd, M. Salganicoff, V. Anand, U. Shreter, S. Vastagh, B. Y. Croft, L. P. Clarke. The lung image database consortium (LIDC) and image database resource initiative (IDRI): A completed reference database of lung nodules on CT scans. Medical Physics, vol.38, no. 2, pp. 915-931, 2011. DOI: 10.1118/1.3528204.

- S. G. Armato III, G. McLennan, L. Bidaut, M. F. McNitt-[152]Gray, C. R. Meyer, A. P. Reeves, B. Zhao, D. R. Aberle, C. I. Henschke, E. A. Hoffman, E. A. Kazerooni, H. MacMahon, E. J. R. Van Beek, D. Yankelevitz, A. M. Biancardi, P. H. Bland, M. S. Brown, R. M. Engelmann, G. E. Laderach, D. Max, R. C. Pais, D. P. Y. Qing, R. Y. Roberts, A. R. Smith, A. Starkey, P. Batra, P. Caligiuri, A. Farooqi, G. W. Gladish, C. M. Jude, R. F. Munden, I. Petkovska, L. E. Quint, L. H. Schwartz, B. Sundaram, L. E. Dodd, C. Fenimore, D. Gur, N. Petrick, J. Freymann, J. Kirby, B. Hughes, A. V. Casteele, S. Gupte, M. Sallam, M. D. Heath, M. H. Kuhn, E. Dharaiya, R. Burns, D. S., Fryd, M. Salganicoff, V. Anand, U. Shreter, S. Vastagh, B. Y. Croft, Clarke, L. P. Data From LIDC-ID-RI [Data set]. The Cancer Imaging Archive. [Online], Available: https://doi.org/10.7937/K9/TCIA.2015.LO 9QL9SX, 2015.
- [153] P. F. Christ, F. Ettlinger, F. Grün, M. E. A. Elshaera, J. Lipkova, S. Schlecht, F. Ahmaddy, S. Tatavarty, M. Bickel, P. Bilic, M. Rempfler, F. Hofmann, M. D. Anastasi, S. A. Ahmadi, G. Kaissis, J. Holch, W. Sommer, R. Braren, V. Heinemann, B. Menze. Automatic liver and tumor segmentation of CT and MRI volumes using cascaded fully convolutional neural networks. [Online], Available: https://arxiv.org/abs/1702.05970, 2017.
- [154] O. Ronneberger, P. Fischer, T. Brox. U-Net: Convolutional networks for biomedical image segmentation. In Proceedings of the 18th International Conference on Medical Image Computing and Computer-assisted Intervention, Springer, Munich, Germany, pp.234–241, 2015. DOI: 10.1007/978-3-319-24574-4_28.
- [155] P. H. Conze, A. E. Kavur, E. Cornec-Le Gall, N. S. Gezer, Y. Le Meur, M. A. Selver, F. Rousseau. Abdominal multiorgan segmentation with cascaded convolutional and adversarial deep networks. *Artificial Intelligence in Medicine*, vol. 117, Article number 102109, 2021. DOI: 10. 1016/j.artmed.2021.102109.
- [156] M. J. Li, W. J. Cai, K. Verspoor, S. R. Pan, X. D. Liang, X. J. Chang. Cross-modal clinical graph transformer for ophthalmic report generation. In *Proceedings of IEEE/CVF Conference on Computer Vision and Pattern Recognition*, IEEE, New Orleans, USA, pp. 20656–20665, 2022. DOI: 10.1109/CVPR52688.2022.02000.
- [157] W. Gómez-Flores, W. C. de Albuquerque Pereira. A comparative study of pre-trained convolutional neural networks for semantic segmentation of breast tumors in ultrasound. *Computers in Biology and Medicine*, vol. 126, Article number 104036, 2020. DOI: 10.1016/j.compbiomed.2020.104036.

- [158] H. Piotrzkowska-Wróblewska, K. Dobruch-Sobczak, M. Byra, A. Nowicki. Open access database of raw ultrasonic signals acquired from malignant and benign breast lesions. *Medical Physics*, vol.44, no.11, pp.6105–6109, 2017. DOI: 10.1002/mp.12538.
- [159] A. Hijab, M. A. Rushdi, M. M. Gomaa, A. Eldeib. Breast cancer classification in ultrasound images using transfer learning. In *Proceedings of the 5th International Conference on Advances in Biomedical Engineering*, IEEE, Tripoli, Lebanon, 2019. DOI: 10.1109/ICABME47164. 2019.8940291.
- [160] G. Ayana, K. Dese, S. W. Choe. Transfer learning in breast cancer diagnoses via ultrasound imaging. *Cancers*, vol.13, no.4, Article number 738, 2021. DOI: 10.3390/ cancers13040738.
- [161] G. Ayana, J. Park, J. W. Jeong, S. W. Choe. A novel multistage transfer learning for ultrasound breast cancer image classification. *Diagnostics*, vol. 12, no. 1, Article number 135, 2022. DOI: 10.3390/diagnostics12010135.
- [162] S. Sudharson, P. Kokil. An ensemble of deep neural networks for kidney ultrasound image classification. Computer Methods and Programs in Biomedicine, vol.197, Article number 105709, 2020. DOI: 10.1016/j.cmpb.2020. 105709.
- [163] W. J. Bai, C. Chen, G. Tarroni, J. M. Duan, F. Guitton, S. E. Petersen, Y. K. Guo, P. M. Matthews, D. Rueckert. Self-supervised learning for cardiac MR image segmentation by anatomical position prediction. In *Proceedings of the 22nd International Conference on Medical Image Computing and Computer Assisted Intervention*, Springer, Shenzhen, China, pp. 541–549, 2019. DOI: 10.1007/ 978-3-030-32245-8 60.
- [164] Y. X. Li, J. W. Chen, X. P. Xie, K. Ma, Y. F. Zheng. Selfloop uncertainty: A novel pseudo-label for semi-supervised medical image segmentation. In *Proceedings of the* 23rd International Conference on Medical Image Computing and Computer Assisted Intervention, Springer, Lima, Peru, pp.614–623, 2020. DOI: 0.1007/978-3-030-59710-8 60.
- [165] C. Doersch, A. Gupta, A. A. Efros. Unsupervised visual representation learning by context prediction. In *Pro*ceedings of *IEEE International Conference on Computer* Vision, Santiago, Chile, pp.1422–1430, 2015. DOI: 10. 1109/ICCV.2015.167.
- [166] D. Pathak, P. Krähenbühl, J. Donahue, T. Darrell, A. A. Efros. Context encoders: Feature learning by inpainting. In Proceedings of IEEE Conference on Computer Vision and Pattern Recognition, Las Vegas, USA, pp.2536– 2544, 2016. DOI: 10.1109/CVPR.2016.278.
- [167] L. Chen, P. Bentley, K. Mori, K. Misawa, M. Fujiwara, D. Rueckert. Self-supervised learning for medical image analysis using image context restoration. *Medical Image Analysis*, vol. 58, Article number 101539, 2019. DOI: 10. 1016/j.media.2019.101539.
- [168] X. L. Zhu, J. W. Yao, J. Z. Huang. Deep convolutional neural network for survival analysis with pathological images. In *Proceedings of IEEE International Conference on Bioinformatics and Biomedicine*, Shenzhen, China, pp. 544–547, 2016. DOI: 10.1109/BIBM.2016.7822579.
- [169] X. L. Zhu, J. W. Yao, F. Y. Zhu, J. Z. Huang. WSISA: Making survival prediction from whole slide histopathological images. In Proceedings of IEEE Conference on Computer Vision and Pattern Recognition, Honolulu, USA,

Machine Intelligence Research 20(2), April 2023

- [170] K. A. Tran, O. Kondrashova, A. Bradley, E. D. Williams, J. V. Pearson, N. Waddell. Deep learning in cancer diagnosis, prognosis and treatment selection. *Genome Medicine*, vol. 13, no. 1, Article number 152, 2021. DOI: 10. 1186/S13073-021-00968-X.
- [171] Y. Li, L. Wang, J. Wang, J. P. Ye, C. K. Reddy. Transfer learning for survival analysis via efficient L2, 1-norm regularized cox regression. In *Proceedings of the 16th IEEE International Conference on Data Mining*, IEEE, Barcelona, Spain, pp.231–240, 2016. DOI: 10.1109/ICDM. 2016.0034.
- [172] R. R. Agravat, M. S. Raval. Brain tumor segmentation and survival prediction. In *Proceedings of the 5th International Workshop on Brainlesion: Glioma, Multiple Sclerosis, Stroke and Traumatic Brain Injuries, Springer, Shenzhen, China, pp. 338–348, 2019. DOI: 10.1007/978-3-*030-46640-4_32.
- [173] A. A. A. Setio, A. Traverso, T. De Bel, M. S. N. Berens, C. Van Den Bogaard, P. Cerello, H. Chen, Q. Dou, M. E. Fantacci, B. Geurts, R. Van Den Gugten, P. A. Heng, B. Jansen, M. M. J. De Kaste, V. Kotov, J. Y. H. Lin, J. T. M. C. Manders, A. Söñora-Mengana, J. C. García-Naranjo, E. Papavasileiou, M. Prokop, M. Saletta, C. M. Schaefer-Prokop, E. T. Scholten, L. Scholten, M. M. Snoeren, E. L. Torres, J. Vandemeulebroucke, N. Walasek, G. C. A. Zuidhof, B. Van Ginneken, C. Jacobs. Validation, comparison, and combination of algorithms for automatic detection of pulmonary nodules in computed tomography images: The LUNA16 challenge. *Medical Image Analysis*, vol. 42, pp. 1–13, 2017. DOI: 10.1016/j.media.2017.06.015.
- [174] R. J. Chen, C. K. Chen, Y. C. Li, T. Y. Chen, A. D. Trister, R. G. Krishnan, F. Mahmood. Scaling vision transformers to gigapixel images via hierarchical self-supervised learning. In *Proceedings of IEEE/CVF Conference* on Computer Vision and Pattern Recognition, IEEE, New Orleans, USA, pp.16123–16134, 2022. DOI: 10. 1109/CVPR52688.2022.01567.
- [175] Y. W. Xu, A. Hosny, R. Zeleznik, C. Parmar, T. Coroller, I. Franco, R. H. Mak, H. J. W. L. Aerts. Deep learning predicts lung cancer treatment response from serial medical imaging. *Clinical Cancer Research*, vol.25, no.11, pp.3266–3275, 2019. DOI: 10.1158/1078-0432.CCR-18-2495.
- [176] T. D. Pham. Time-frequency time-space long short-term memory networks for image classification of histopathological tissue. *Scientific Reports*, vol. 11, no. 1, Article number 13703, 2021. DOI: 10.1038/s41598-021-93160-5.
- [177] A. Konwer, X. Xu, J. Bae, C. Chen, P. Prasanna. Temporal context matters: Enhancing single image prediction with disease progression representations. In Proceedings of IEEE/CVF Conference on Computer Vision and Pattern Recognition, IEEE, New Orleans, USA, pp. 18802–18813, 2022. DOI: 10.1109/CVPR52688.2022. 01826.
- [178] R. Q. Gao, Y. K. Huo, S. X. Bao, Y. C. Tang, S. L. Antic, E. S. Epstein, A. B. Balar, S. Deppen, A. B. Paulson, K. L. Sandler, P. P. Massion, B. A. Landman. Distanced LSTM: Time-distanced gates in long short-term memory models for lung cancer detection. In *Proceedings of the* 10th International Workshop on Machine Learning in Medical Imaging, Springer, Shenzhen, China, pp.310– 318, 2019. DOI: 10.1007/978-3-030-32692-0_36.
- 🖗 Springer

- [179] J. Ouyang, Q. Y. Zhao, E. Adeli, E. V. Sullivan, A. Pfefferbaum, G. Zaharchuk, K. M. Pohl. Self-supervised longitudinal neighbourhood embedding. In Proceedings of the 24th International Conference on Medical Image Computing and Computer Assisted Intervention, Springer, Strasbourg, France, pp. 80–89, 2021. DOI: 10.1007/ 978-3-030-87196-3_8.
- [180] K. Antczak. Deep recurrent neural networks for ECG signal denoising. [Online], Available: https://arxiv.org/abs/ 1807.11551, 2018.
- [181] K. Antczak. A generative adversarial approach to ECG synthesis and denoising. [Online], Available: https://arxiv.org/abs/2009.02700, 2020.
- [182] R. Rodrigues, P. Couto. Semi-supervised learning for ECG classification. In *Proceedings of Computing in Cardiology*, IEEE, Brno, Czech Republic, 2021. DOI: 10. 23919/CinC53138.2021.9662693.
- [183] J. H. Jang, T. Y. Kim, D. Yoon. Effectiveness of transfer learning for deep learning-based electrocardiogram analysis. *Healthcare Informatics Research*, vol.27, no.1, pp.19–28, 2021. DOI: 10.4258/hir.2021.27.1.19.
- [184] M. T. Almalchy, S. M. S. ALGayar, N. Popescu. Atrial fibrillation automatic diagnosis based on ECG signal using pretrained deep convolution neural network and SVM multiclass model. In *Proceedings of the 13th International Conference on Communications*, IEEE, Bucharest, Romania, pp.197–202, 2020. DOI: 10.1109/COMM48946. 2020.9141994.
- [185] A. Qayyum, F. Mériaudeau, G. C. Y. Chan. Classification of atrial fibrillation with pre-trained convolutional neural network models. In *Proceedings of IEEE/EMBS Conference on Biomedical Engineering and Sciences*, IEEE, Sarawak, Malaysia, pp.594–599, 2018. DOI: 10. 1109/IECBES.2018.8626624.
- [186] D. Kiyasseh, T. T. Zhu, D. A. Clifton. CLOCS: Contrastive learning of cardiac signals across space, time, and patients. In *Proceedings of the 38th International Conference on Machine Learning*, pp. 5606–5615, 2021.
- [187] D. Gedon, A. H. Ribeiro, N. Wahlström, T. B. Schön. First steps towards self-supervised pretraining of the 12lead ECG. In *Proceedings of Computing in Cardiology*, IEEE, Brno, Czech Republic, 2021. DOI: 10.23919/ CinC53138.2021.9662748.
- [188] T. Mehari, N. Strodthoff. Self-supervised representation learning from 12-lead ECG data. Computers in Biology and Medicine, vol.141, Article number 105114, 2022. DOI: 10.1016/j.compbiomed.2021.105114.
- [189] H. Liu, Z. B. Zhao, Q. She. Self-supervised ECG pretraining. *Biomedical Signal Processing and Control*, vol.70, Article number 103010, 2021. DOI: 10.1016/j. bspc.2021.103010.
- [190] J. Y. Cheng, H. Goh, K. Dogrusoz, O. Tuzel, E. Azemi. Subject-aware contrastive learning for biosignals. [Online], Available: https://arxiv.org/abs/2007.04871, 2020.
- [191] X. Zhang, Z. Y. Zhao, T. Tsiligkaridis, M. Zitnik. Self-supervised contrastive pre-training for time series via timefrequency consistency. [Online], Available: https://arxiv. org/abs2206.08496, 2022.
- [192] H. Chen, G. J. Wang, G. D. Zhang, P. Zhang, H. Z. Yang. CLECG: A novel contrastive learning framework for electrocardiogram arrhythmia classification. *IEEE Signal Processing Letters*, vol. 28, pp. 1993–1997, 2021. DOI: 10.

1109/LSP.2021.3114119.

- [193] K. Radhika, V. R. M. Oruganti. Transfer learning for subject-independent stress detection using physiological signals. In *Proceedings of the 17th IEEE India Council International Conference*, New Delhi, India, 2020. DOI: 10.1109/INDICON49873.2020.9342505.
- [194] P. Sarkar, S. Lobmaier, B. Fabre, D. González, A. Mueller, M. G. Frasch, M. C. Antonelli, A. Etemad. Detection of maternal and fetal stress from the electrocardiogram with self-supervised representation learning. *Scientific Reports*, vol. 11, no. 1, Article number 24146, 2021. DOI: 10.1038/S41598-021-03376-8.
- [195] P. Sarkar, A. Etemad. Self-supervised ECG representation learning for emotion recognition. *IEEE Transactions* on Affective Computing, vol. 13, no.3, pp.1541–1554, 2022. DOI: 10.1109/TAFFC.2020.3014842.
- [196] P. J. Aston, J. V. Lyle, E. Bonet-Luz, C. L. Huang, Y. M. Zhang, K. Jeevaratnam, M. Nandi. Deep learning applied to attractor images derived from ECG signals for detection of genetic mutation. In *Proceedings of Computing in Cardiology*, IEEE, Singapore, 2019. DOI: 10. 22489/CinC.2019.097.
- [197] Y. Cimtay, E. Ekmekcioglu. Investigating the use of pretrained convolutional neural network on cross-subject and cross-dataset EEG emotion recognition. *Sensors*, vol. 20, no. 7, Article number 2034, 2020. DOI: 10.3390/ s20072034.
- [198] S. Bagherzadeh, K. Maghooli, A. Shalbaf, A. Maghsoudi. Recognition of emotional states using frequency effective connectivity maps through transfer learning approach from electroencephalogram signals. *Biomedical Signal Processing and Control*, vol. 75, Article number 103544, 2022. DOI: 10.1016/j.bspc.2022.103544.
- [199] M. N. Mohsenvand, M. R. Izadi, P. Maes. Contrastive representation learning for electroencephalogram classification. In *Proceedings of the Machine Learning for Health*, pp. 238–253, 2020.
- [200] S. Raghu, N. Sriraam, Y. Temel, S. V. Rao, P. L. Kubben. EEG based multi-class seizure type classification using convolutional neural network and transfer learning. *PMLR Neural Networks*, vol.124, pp.202–212, 2020. DOI: 10.1016/j.neunet.2020.01.017.
- [201] H. S. Nogay, H. Adeli. Detection of epileptic seizure using pretrained deep convolutional neural network and transfer learning. *European Neurology*, vol.83, no.6, pp. 602–614, 2020. DOI: 10.1159/000512985.
- [202] S. Y. Tang, J. Dunnmon, K. K. Saab, X. Zhang, Q. Y. Huang, F. Dubost, D. Rubin, C. Lee-Messer. Self-supervised graph neural networks for improved electroencephalographic seizure analysis. In *Proceedings of the 10th International Conference on Learning Representations*, 2022.
- [203] J. J. Xu, Y. J. Zheng, Y. F. Mao, R. X. Wang, W. S. Zheng. Anomaly detection on electroencephalography with self-supervised learning. In *Proceedings of IEEE International Conference on Bioinformatics and Biomedicine*, Seoul, Republic of Korea, pp.363–368, 2020. DOI: 10.1109/BIBM49941.2020.9313163.
- [204] H. Banville, O. Chehab, A. Hyvärinen, D. A. Engemann, A. Gramfort. Uncovering the structure of clinical EEG signals with self-supervised learning. *Journal of Neural Engineering*, vol. 18, no. 4, Article number 046020, 2021. DOI: 10.1088/1741-2552/abca18.

- [205] M. T. Sadiq, M. Z. Aziz, A. Almogren, A. Yousaf, S. Siuly, A. U. Rehman. Exploiting pretrained CNN models for the development of an EEG-based robust BCI framework. *Computers in Biology and Medicine*, vol. 143, Art-
- [206] Y. H. Ou, S. Q. Sun, H. T. Gan, R. Zhou, Z. Yang. An improved self-supervised learning for EEG classification. *Mathematical Biosciences and Engineering*, vol. 19, no.7, pp. 6907–6922, 2022. DOI: 10.3934/mbe.2022325.

2022.105242.

icle number 105242, 2022. DOI: 10.1016/j.compbiomed.

- [207] H. Phan, O. Y. Chén, P. Koch, Z. Q. Lu, I. McLoughlin, A. Mertins, M. De Vos. Towards more accurate automatic sleep staging via deep transfer learning. *IEEE Transactions on Biomedical Engineering*, vol.68, no.6, pp. 1787–1798, 2021. DOI: 10.1109/TBME.2020.3020381.
- [208] X. Jiang, J. H. Zhao, B. Du, Z. Y. Yuan. Self-supervised contrastive learning for EEG-based sleep staging. In Proceedings of International Joint Conference on Neural Networks, IEEE, Shenzhen, China, 2021. DOI: 10.1109/ IJCNN52387.2021.9533305.
- [209] N. Wagh, J. H. Wei, S. Rawal, B. M. Berry, L. Barnard, B. Brinkmann, G. Worrell, D. Jones, Y. Varatharajah. Domain-guided self-supervision of EEG data improves downstream classification performance and generalizability. In *Proceedings of Machine Learning for Health*, pp. 130–142, 2021.
- [210] R. W. Picard, E. Vyzas, J. Healey. Toward machine emotional intelligence: Analysis of affective physiological state. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, vol.23, no.10, pp.1175–1191, 2001. DOI: 10.1109/34.954607.
- [211] D. R. Simkin, R. W. Thatcher, J. Lubar. Quantitative EEG and neurofeedback in children and adolescents: Anxiety disorders, depressive disorders, comorbid addiction and attention-deficit/hyperactivity disorder, and brain injury. Child and Adolescent Psychiatric Clinics of North America, vol. 23, no. 3, pp. 427–464, 2014. DOI: 10. 1016/j.chc.2014.03.001.
- [212] G. Z. Zhao, Y. Ge, B. Y. Shen, X. J. Wei, H. Wang. Emotion analysis for personality inference from EEG signals. *IEEE Transactions on Affective Computing*, vol.9, no.3, pp. 362–371, 2017. DOI: 10.1109/TAFFC.2017.2786207.
- [213] N. Lu, T. F. Li, X. D. Ren, H. Y. Miao. A deep learning scheme for motor imagery classification based on restricted boltzmann machines. *IEEE Transactions on Neural Systems and Rehabilitation Engineering*, vol. 25, no.6, pp. 566–576, 2017. DOI: 10.1109/TNSRE.2016.2601240.
- [214] M. McDermott, B. Nestor, E. Kim, W. C. Zhang, A. Goldenberg, P. Szolovits, M. Ghassemi. A comprehensive EHR timeseries pre-training benchmark. In *Proceedings of the Conference on Health, Inference, and Learning*, ACM, pp.257–278, 2021. DOI: 10.1145/3450439. 3451877.
- [215] H. Chen, S. M. Lundberg, G. Erion, J. H. Kim, S. I. Lee. Forecasting adverse surgical events using self-supervised transfer learning for physiological signals. *Digital Medicine*, vol.4, no.1, Article number 167, 2021. DOI: 10. 1038/s41746-021-00536-y.
- [216] X. Xu, X. Xu, Y. Y. Sun, X. S. Liu, X. Li, G. T. Xie, F. Wang. Predictive modeling of clinical events with mutual enhancement between longitudinal patient records and medical knowledge graph. In *Proceedings of IEEE International Conference on Data Mining*, Auckland, New

Zealand, pp.777–786, 2021. DOI: 10.1109/ICDM51629. 2021.00089.

- [217] Y. Xue, N. Du, A. Mottram, M. Seneviratne, A. M. Dai. Learning to select best forecast tasks for clinical outcome prediction. In *Proceedings of the 34th Conference on Neural Information Processing Systems*, Vancouver, Canada, pp.15031–15041, 2020.
- [218] S. Tipirneni, C. K. Reddy. Self-supervised transformer for sparse and irregularly sampled multivariate clinical timeseries. ACM Transactions on Knowledge Discovery from Data, vol.1, no.1, Article number 105, 2022. DOI: 10. 1145/3516367.
- [219] H. X. Ren, J. Y. Wang, W. X. Zhao, N. Wu. RAPT: Pretraining of time-aware transformer for learning robust healthcare representation. In *Proceedings of the 27th* ACM SIGKDD Conference on Knowledge Discovery & Data Mining, Singapore, pp.3503–3511, 2021. DOI: 10. 1145/3447548.3467069.
- [220] B. van Aken, J. M. Papaioannou, M. Mayrdorfer, K. Budde, F. Gers, A. Löser. Clinical outcome prediction from admission notes using self-supervised knowledge integration. In *Proceedings of the 16th Conference of the European Chapter of the Association for Computational Linguistics*, pp.881–893, 2021. DOI: 10.18653/v1/2021. eacl-main.75.
- [221] C. Lu, C. K. Reddy, Y. Ning. Self-supervised graph learning with hyperbolic embedding for temporal health event prediction. *IEEE Transactions on Cybernetics*, to be published. DOI: 10.1109/TCYB.2021.3109881.
- [222] K. Hur, J. Lee, J. Oh, W. Price, Y. Kim, E. Choi. Unifying heterogeneous electronic health records systems via text-based code embedding. In *Proceedings of Conference on Health, Inference, and Learning*, pp.183–203, 2022.
- [223] S. Biswal, C. Xiao, L. M. Glass, E. Milkovits, J. M. Sun. Doctor2Vec: Dynamic doctor representation learning for clinical trial recruitment. *Proceedings of AAAI Conference on Artificial Intelligence*, vol.34, no.1, pp.557– 564, 2020. DOI: 10.1609/aaai.v34i01.5394.
- [224] Y. P. Chen, Y. H. Lo, F. P. Lai, C. H. Huang. Disease concept-embedding based on the self-supervised method for medical information extraction from electronic health records and disease retrieval: Algorithm development and validation study. *Journal of Medical Internet Research*, vol. 23, no. 1, Article number e25113, 2021. DOI: 10.2196/ 25113.
- [225] E. Lehman, S. Jain, K. Pichotta, Y. Goldberg, B. C. Wallace. Does BERT pretrained on clinical notes reveal sensitive data? In Proceedings of Conference of the North American Chapter of the Association for Computational Linguistics: Human Language Technologies, pp.946–959, 2021. DOI: 10.18653/v1/2021.naacl-main.73.
- [226] X. Y. Zhang, C. Xiao, L. M. Glass, J. M. Sun. DeepEnroll: Patient-trial matching with deep embedding and entailment prediction. In *Proceedings of The Web Conference*, ACM, Taipei, China, pp. 1029–1037, 2020. DOI: 10. 1145/3366423.3380181.
- [227] H. D. Hlynsson, S. Ellertsson, J. F. Daðason, E. L. Sigurdsson, H. Loftsson. Semi-self-supervised automated ICD coding. [Online], Available: https://arxiv.org/abs/ 2205.10088, 2022.
- [228] Z. Zhang, J. S. Liu, N. Razavian. BERT-XML: Large

scale automated ICD coding using BERT pretraining. In Proceedings of the 3rd Clinical Natural Language Processing Workshop, pp.24–34, 2020. DOI: 10.18653/v1/ 2020.clinicalnlp-1.3.

- [229] Y. Q. Su, Y. L. Shi, W. Lee, L. Cheng, H. M. Guo. TAHDNet: Time-aware hierarchical dependency network for medication recommendation. *Journal of Biomedical Informatics*, vol.129, Article number 104069, 2022. DOI: 10.1016/j.jbi.2022.104069.
- [230] J. Y. Shang, T. F. Ma, C. Xiao, J. M. Sun. Pre-training of graph augmented transformers for medication recommendation. In *Proceedings of the 28th International Joint Conference on Artificial Intelligence*, Macao, China, pp. 5953–5959, 2019. DOI: 10.24963/ijcai.2019/825.
- [231] Z. Sun, S. L. Peng, Y. N. Yang, X. Q. Wang, F. Li. A general fine-tuned transfer learning model for predicting clinical task acrossing diverse EHRs datasets. In *Proceedings* of *IEEE International Conference on Bioinformatics and Biomedicine*, San Diego, USA, pp.490–495, 2019. DOI: 10.1109/BIBM47256.2019.8983098.
- [232] L. T. Ma, X. Y. Ma, J. Y. Gao, X. F. Jiao, Z. H. Yu, C. H. Zhang, W. J. Ruan, Y. S. Wang, W. Tang, J. T. Wang. Distilling knowledge from publicly available online EMR data to emerging epidemic for prognosis. In *Proceedings* of Web Conference, ACM, Ljubljana, Slovenia, pp. 3558– 3568, 2021. DOI: 10.1145/3442381.3449855.
- [233] H. Quan, V. Sundararajan, P. Halfon, A. Fong, B. Burnand, J. C. Luthi, L. D. Saunders, C. A. Beck, T. E. Feasby, W. A. Ghali. Coding algorithms for defining comorbidities in ICD-9-CM and ICD-10 administrative data. *Medical Care*, vol. 43, no. 11, pp. 1130–1139, 2005. DOI: 10.1097/01.mlr.0000182534.19832.83.
- [234] Y. Y. Zhang, X. Wu, Q. Fang, S. S. Qian, C. S. Xu. Knowledge-enhanced attributed multi-task learning for medicine recommendation. ACM Transactions on Information Systems, to be published. DOI: 10.1145/ 3527662.
- [235] Y. K. Li, H. Y. Wang, Y. Luo. A comparison of pretrained vision-and-language models for multimodal representation learning across medical images and reports. In Proceedings of IEEE International Conference on Bioinformatics and Biomedicine, Seoul, Republic of Korea, pp.1999–2004, 2020. DOI: 10.1109/BIBM49941. 2020.9313289.
- [236] L. H. Li, M. Yatskar, D. Yin, C. J. Hsieh, K. W. Chang. VisualBERT: A simple and performant baseline for vision and language. [Online], Available: https://arxiv.org/ abs/1908.03557, 2019.
- [237] Y. C. Chen, L. J. Li, L. C. Yu, A. El Kholy, F. Ahmed, Z. Gan, Y. Cheng, J. J. Liu. UNITER: UNiversal Image-TExt representation learning. [Online], Available: https:// arxiv.org/abs/1909.11740, 2019.
- [238] H. Tan, M. Bansal. LXMERT: Learning cross-modality encoder representations from transformers. In Proceedings of Conference on Empirical Methods in Natural Language Processing and the 9th International Joint Conference on Natural Language Processing, Hong Kong, China, pp. 5100–5111, 2019. DOI: 10.18653/v1/D19-1514.
- [239] Z. C. Huang, Z. Y. Zeng, B. Liu, D. M. Fu, J. L. Fu. Pixel-BERT: Aligning image pixels with text by deep multi-modal transformers. [Online], Available: https:// arxiv.org/abs/2004.00849, 2020.

🖉 Springer

- [240] Y. Khare, V. Bagal, M. Mathew, A. Devi, U. D. Priyakumar, C. V. Jawahar. MMBERT: Multimodal BERT pretraining for improved medical VQA. In Proceedings of the 18th IEEE International Symposium on Biomedical Imaging, Nice, France, pp.1033-1036, 2021. DOI: 10. 1109/ISBI48211.2021.9434063.
- [241] N. Rieke, J. Hancox, W. Q. Li, F. Milletari, H. R. Roth, S. Albarqouni, S. Bakas, M. N. Galtier, B. A. Landman, K. Maier-Hein, S. Ourselin, M. Sheller, R. M. Summers, A. Trask, D. G. Xu, M. Baust, M. J. Cardoso. The future of digital health with federated learning. Digital Medicine, vol.3, no.1, Article number 119, 2020. DOI: 10.1038/ s41746-020-00323-1.
- [242] M. Abadi, A. Chu, I. Goodfellow, H. B. McMahan, I. Mironov, K. Talwar, L. Zhang. Deep learning with differential privacy. In Proceedings of ACM SIGSAC Conference on Computer and Communications Security, Vienna, Austria, pp. 308-318, 2016. DOI: 10.1145/2976749. 2978318.



Yixuan Qiu received the M.Sc. degree in electrical engineering from The University of Queensland, Australia in 2020. Currently, he is a Ph.D. degree candidate in data science at School of information Technology and Electrical Engineering, The University of Queensland, Australia.

His research interests include medical data analytic, self-supervised learning and

federated learning. E-mail: y.qiu@uq.edu.au RCID iD: 0000-0002-7593-1876





Feng Lin received the M.Sc. degree from The University of Queensland, Australia in 2022. He is currently working at Wipro, Australia

His research interests include weaklysupervised learning, data mining and deep learning.

Weitong Chen received the Ph.D. de-

gree in computer science from The University of Queensland, Australia in 2020.

He is currently a lecturer at The Uni-

learning and its application to medical do-

His research interests include machine

E-mail: feng.lin@uq.net.au

versity of Adelaide, Australia.



mains. E-mail: t.chen@adelaide.edu.au (Cor-

responding author) ORCID iD: 0000-0003-1001-7925



Miao Xu received the Ph.D. degree in machine learning from Nanjing University, China in 2017. She is a lecturer at The University of Queensland (UQ), Australia. Before joining UQ, she was a postdoctoral researcher at RIKEN, Japan.

Her research interests include weakly supervised learning and its application to the medical and cyber-security domains.

E-mail: miao.xu@uq.edu.au